0288-A

Frace Relations and Equal Education Opportunities at Provise West High School

23. S. Commission on Civil Rights

March 1996



The Unit 1 Lies Commission on Chill Alights

The United Stains Commission on Civil Rights, first scenaes by the Civil hights Act of 1983, is an independent, higher land agency of the Federal Government. By the term, of the 1983 act as amended by the Chil Rights Commission Assessments Act of 1984, the Commission is charged with the following duties pertaining to discrimination or decisis of the equal protection of the laws based on race, color, religion, sex, age, discribility, or national origin, at it, the administration of justice investigation of individual discriminatory decisals of the equal protection of the law; appraisable of the laws and policies of the United States with respect to discrimination or decisis of equal protection of the law; amintaneous of a national classinglessor for information or decisis of equal protection of the law; amintaneous of a national classinglessor for information respecting discrimination or decisis of equal protection of the law; amintaneous of a national classinglessor for information respecting discrimination or decisis of equal protection of the law investigation of the law investigation of the law; amintaneous of federal electronic and propagates and propagates and propagates and insulation of the provident and the Congress, or the President shall deem designates.

The State Advisory Committees

An Advancy Committee — one United States Commission on Civil Rights has been established in an other 50 States and the District of Ordenbin provenent to section 100(s) of the Civil and Act of 1307 and section 3(d) of the Civil Rights Commission Amendments Act of 1964, 20 and victory Commission are made up of Representate parameters in the Civil Rights Commission of the Civil Rights of the Commission of the Commission of the Commission of the Civil Rights o

Race Relations and Equal Education Opportunities at Proviso West High School

Illinois Advisory Committee to the U. S. Commission on Civil Rights

March 1996

A report of the Illinois Advisory Committee to the United States Commission on Civil Rights prepared for the information and consideration of the Commission. This report will be considered by the Commission and the Commission will make public its reaction.

Letter of Transmittal

Illinois Advisory Committee to the U.S. Commission on Civil Rights

Members of the Commission
Mary Frances Berry, Chairperson
Cruz Reynoso, Vice Chairperson
Carl A. Anderson
Robert P. George
A. Leon Higginbotham, Jr.
Constance Horner
Yvonne Y. Lee
Russell G. Redenbaugh

Mary K. Mathews, Staff Director

The Illinois Advisory Committee submits this report, Race Relations and Equal Education Opportunities at Provise West High School, as part of its responsibility to advise the Commission on civil rights issues within the State. The report was unanimously adopted by the Advisory Committee by a 12–0 vote. The Advisory Committee is indebted to the Midwestern Regional Office for its assistance in organizing the forum and preparing this report.

The Advisory Committee held a community forum at Proviso West High School on November 30, 1994, and December 1, 1994. The meeting was held subsequent to an article in the New York Times Magazine on May 25, 1994, by H.G. Bissinger, "We're All Racist Now." At the request of the Commission, we solicited student, faculty, and community opinion on the racial climate and educational opportunities at Proviso West high school.

The Advisory Committee heard from 45 students, 18 faculty and staff, administrators, parents, and individuals who live in the surrounding community. To obtain an accurate reflection of student opinion, a sample of 30 students was randomly drawn from the student population; additionally, another group of 15 student leaders was interviewed. Fifteen of the 18 faculty participants were also selected at random. In addition, students and faculty quoted in the article were afforded the opportunity to speak at the meeting. There was also a public session, at which anyone could address the Advisory Committee, and written comments were accepted.

Additionally, the author of the article, H.G. Bissinger, was provided a draft of the report. His response is in appendix V.

The Advisory Committee hopes the Commission finds this report of value in its monitoring of racial and ethnic tensions.

seal D. Marhaman

Respectfully,

Joseph D. Mathewson, Chairperson

Illinois Advisory Committee

Illinois Advisory Committee to the U.S. Commission on Civil Rights

Joseph D. Mathewson, Chairperson Winnetka

Rose Mary Bombela Chicago

Nancy S. Chen Naperville

Jae K. Choi† Chicago

Preston E. Ewing Cairo

CM. 0

Janie Khoury† Chicago

Faye Lyon* Winnetka

Connie Peters
Arlington Heights

J. Thomas Pugh Peoria

Dorris M. Roberts Chicago

James E. Scales Carbondale

Hugh J. Schwartzberg Chicago

Kenneth B. Smith

Chicago

Staci M. Yandle Belleville



^{*} Faye Lyon was Chairperson of the Advisory Committee at the time the community forum took place.

[†] Was not a member of the Advisory Committee at the time the community forum took place.

Contents

Part 2. Proviso West High School	5
Student Background, Enrollment, and Achievement	
Teacher Background, Tenure, and Class Size	
School Discipline	
Part 3. Student Comments	11
Selected Student Leaders	
Students Selected by Random Sample	
Volunteer Student Comments	
Volument Comments	20
Part 4. Faculty and Staff Comments	20
Administration	20
Faculty and Staff	22
Part 5. Parent and Community Comments	
Parents	
Community	33
Part 6. Conclusions and Observations	38
Tables 1. Population by Race/Ethnicity of Communities in Proviso Township High School District 20	
0 Davial/RAN '- Davida 1 D. Ara 1 M.A. D	K
2. Racial/Ethnic Background Rates and Total Enrollment	
3. Limited-English Proficient and Low-Income Student Rates	6
Limited-English Proficient and Low-Income Student Rates	6
3. Limited-English Proficient and Low-Income Student Rates	6 7
3. Limited-English Proficient and Low-Income Student Rates 4. Attendance Rates, Dropout Rates, and Chronic Truancy 5. Graduation Rates 6. IGAP Average Scores for Reading, Mathematics, Writing, Science, and Social Studies	677
3. Limited-English Proficient and Low-Income Student Rates 4. Attendance Rates, Dropout Rates, and Chronic Truancy 5. Graduation Rates 6. IGAP Average Scores for Reading, Mathematics, Writing, Science, and Social Studies 7. American College Testing Scores, 1993–94	6 7 7
3. Limited-English Proficient and Low-Income Student Rates 4. Attendance Rates, Dropout Rates, and Chronic Truancy 5. Graduation Rates 6. IGAP Average Scores for Reading, Mathematics, Writing, Science, and Social Studies 7. American College Testing Scores, 1993–94 8. Teacher Racial/Ethnic Background	6 7 7 8 9
3. Limited-English Proficient and Low-Income Student Rates 4. Attendance Rates, Dropout Rates, and Chronic Truancy 5. Graduation Rates 6. IGAP Average Scores for Reading, Mathematics, Writing, Science, and Social Studies 7. American College Testing Scores, 1993–94 8. Teacher Racial/Ethnic Background 9. Mean Teacher Experience	6 7 7 8 9
3. Limited-English Proficient and Low-Income Student Rates 4. Attendance Rates, Dropout Rates, and Chronic Truancy 5. Graduation Rates 6. IGAP Average Scores for Reading, Mathematics, Writing, Science, and Social Studies 7. American College Testing Scores, 1993–94 8. Teacher Racial/Ethnic Background 9. Mean Teacher Experience 10. Teacher Salary and Per Pupil Cost	6 7 7 8 9 9
3. Limited-English Proficient and Low-Income Student Rates 4. Attendance Rates, Dropout Rates, and Chronic Truancy 5. Graduation Rates 6. IGAP Average Scores for Reading, Mathematics, Writing, Science, and Social Studies 7. American College Testing Scores, 1993–94 8. Teacher Racial/Ethnic Background 9. Mean Teacher Experience 10. Teacher Salary and Per Pupil Cost 11. Class Size	677899
3. Limited-English Proficient and Low-Income Student Rates 4. Attendance Rates, Dropout Rates, and Chronic Truancy 5. Graduation Rates 6. IGAP Average Scores for Reading, Mathematics, Writing, Science, and Social Studies 7. American College Testing Scores, 1993–94 8. Teacher Racial/Ethnic Background 9. Mean Teacher Experience 10. Teacher Salary and Per Pupil Cost 11. Class Size 12. School Suspensions by Race and Ethnicity	6 7 7 8 9 9
3. Limited-English Proficient and Low-Income Student Rates 4. Attendance Rates, Dropout Rates, and Chronic Truancy 5. Graduation Rates 6. IGAP Average Scores for Reading, Mathematics, Writing, Science, and Social Studies 7. American College Testing Scores, 1993–94 8. Teacher Racial/Ethnic Background 9. Mean Teacher Experience 10. Teacher Salary and Per Pupil Cost 11. Class Size 12. School Suspensions by Race and Ethnicity 13. Student Leader Participants	6 7 7 8 9 9 9
3. Limited-English Proficient and Low-Income Student Rates 4. Attendance Rates, Dropout Rates, and Chronic Truancy 5. Graduation Rates 6. IGAP Average Scores for Reading, Mathematics, Writing, Science, and Social Studies 7. American College Testing Scores, 1993–94 8. Teacher Racial/Ethnic Background 9. Mean Teacher Experience 10. Teacher Salary and Per Pupil Cost 11. Class Size 12. School Suspensions by Race and Ethnicity 13. Student Leader Participants 14. Student Participants	6 7 7 8 9 9 9 10
3. Limited-English Proficient and Low-Income Student Rates 4. Attendance Rates, Dropout Rates, and Chronic Truancy 5. Graduation Rates 6. IGAP Average Scores for Reading, Mathematics, Writing, Science, and Social Studies 7. American College Testing Scores, 1993–94 8. Teacher Racial/Ethnic Background 9. Mean Teacher Experience 10. Teacher Salary and Per Pupil Cost 11. Class Size 12. School Suspensions by Race and Ethnicity 13. Student Leader Participants 14. Student Participants 15. Faculty and Staff Participants	6 7 7 8 9 9 9 10 10
3. Limited-English Proficient and Low-Income Student Rates 4. Attendance Rates, Dropout Rates, and Chronic Truancy 5. Graduation Rates 6. IGAP Average Scores for Reading, Mathematics, Writing, Science, and Social Studies 7. American College Testing Scores, 1993–94 8. Teacher Racial/Ethnic Background 9. Mean Teacher Experience 10. Teacher Salary and Per Pupil Cost 11. Class Size 12. School Suspensions by Race and Ethnicity 13. Student Leader Participants 14. Student Participants	6 7 7 8 9 9 9 10 10
3. Limited-English Proficient and Low-Income Student Rates 4. Attendance Rates, Dropout Rates, and Chronic Truancy 5. Graduation Rates 6. IGAP Average Scores for Reading, Mathematics, Writing, Science, and Social Studies 7. American College Testing Scores, 1993–94 8. Teacher Racial/Ethnic Background 9. Mean Teacher Experience 10. Teacher Salary and Per Pupil Cost 11. Class Size 12. School Suspensions by Race and Ethnicity 13. Student Leader Participants 14. Student Participants 15. Faculty and Staff Participants	6 7 7 8 9 9 9 10 10
3. Limited-English Proficient and Low-Income Student Rates 4. Attendance Rates, Dropout Rates, and Chronic Truancy 5. Graduation Rates 6. IGAP Average Scores for Reading, Mathematics, Writing, Science, and Social Studies 7. American College Testing Scores, 1993–94 8. Teacher Racial/Ethnic Background 9. Mean Teacher Experience 10. Teacher Salary and Per Pupil Cost 11. Class Size 12. School Suspensions by Race and Ethnicity 13. Student Leader Participants 14. Student Participants 15. Faculty and Staff Participants 16. Parent and Community Participants Appendices	67899101012152228
3. Limited-English Proficient and Low-Income Student Rates 4. Attendance Rates, Dropout Rates, and Chronic Truancy 5. Graduation Rates 6. IGAP Average Scores for Reading, Mathematics, Writing, Science, and Social Studies 7. American College Testing Scores, 1993—94 8. Teacher Racial/Ethnic Background 9. Mean Teacher Experience 10. Teacher Salary and Per Pupil Cost 11. Class Size 12. School Suspensions by Race and Ethnicity 13. Student Leader Participants 14. Student Participants 15. Faculty and Staff Participants 16. Parent and Community Participants 16. Parent and Community Participants I. New York Times Magazine article, "We're All Racist Now"	677899101012152228
3. Limited-English Proficient and Low-Income Student Rates 4. Attendance Rates, Dropout Rates, and Chronic Truancy 5. Graduation Rates 6. IGAP Average Scores for Reading, Mathematics, Writing, Science, and Social Studies 7. American College Testing Scores, 1993—94 8. Teacher Racial/Ethnic Background 9. Mean Teacher Experience 10. Teacher Salary and Per Pupil Cost 11. Class Size 12. School Suspensions by Race and Ethnicity 13. Student Leader Participants 14. Student Participants 15. Faculty and Staff Participants 16. Parent and Community Participants 17. New York Times Magazine article, "We're All Racist Now" 18. Letters from the Commission to the Attorney General and the Secretary of Education	678991012152228
3. Limited-English Proficient and Low-Income Student Rates 4. Attendance Rates, Dropout Rates, and Chronic Truancy 5. Graduation Rates 6. IGAP Average Scores for Reading, Mathematics, Writing, Science, and Social Studies 7. American College Testing Scores, 1993—94 8. Teacher Racial/Ethnic Background 9. Mean Teacher Experience 10. Teacher Salary and Per Pupil Cost 11. Class Size 12. School Suspensions by Race and Ethnicity 13. Student Leader Participants 14. Student Participants 15. Faculty and Staff Participants 16. Parent and Community Participants 16. Parent and Community Participants I. New York Times Magazine article, "We're All Racist Now"	6778991012152228



Part 1

Introduction

The cover story of the May 29, 1994, New York
Times Magazine reported on Proviso West
High School, a public high school in Hillside,
Illinois. The school is located in a near western
suburb of Chicago near the juncture of the Eisenhower expressway and the tristate tollway. The
author, H.G. Bissinger, writes:

Proviso West started as a jewel of a suburban high school. But then for 20 years the school board, faced with the needs of a growing influx of black students, fretted over maintenance and patronage hiring. Whites fled, standards fell and "We're all racist now."

Proviso West High School is one of two high schools in Proviso Township High School District 209; the other is Proviso East High School. Ten communities feed the high school district: Bellwood, Berkeley, Broadview, Forest Park, Hillside, Maywood, Melrose Park, Northlake, Stone Park, and Westchester. Table 1 gives the population and racial breakdown of the communities.

The Times article described Proviso West as a school built during the 1950s in a then all-white suburban area west of Chicago. At the time of its building, it was considered a paragon of educational opportunity for its students. But conditions in and around the school changed. Between 1970 and 1990 the racial composition of the school became predominantly minority, and residents of the community voted down all school referendums.

Built in the late 1950s with a \$6.75 million bond issue, [Proviso West] was more than a school: it was a symbol. At the dedication coromonies on Nov. 16, 1958, the

TABLE (1)	Hamilton and the		Processing with	ng High Silver Destrict 209
	White	Bleck	Latino	Other
Bellwood	4,457	14,240	1,197	347
Berkeley	4,390	230	304	213
Broadview	3,6 9 6	4,631	187	199
Forest Park	10,983	1,926	734	1,275
Hillside	6,366	472	440	394
Maywood	2,631	22,542	1,795	171
Melrose Park	13,884	155	6,303	517
Northlake	NA.	NA.		NA.
Stone Park	1,605	19	2,544	155
Westchester	16,377	144	315	465

¹ H.G. Bissinger, "We're All Racist Now," The New York Times Magazine, May 29, 1994 p. 27 (hereafter referred to as Times article). The complete article is in app. 1.

words of the American educator James Bryant Conant wei ainvoked, "Our purpose is to cultivate in the largest number of our future citizens an appreciation of both the responsibilities and the benefits which come to them because they are American and free."...

Over the past decade, Proviso West has experienced a dizzying degree of demographic and socioeconomic change. In the 1990-91 academic year, 11 percent of the students were from low-income families. This year [1994], the figure has almost doubled, to 19 percent. In 1973-74, the school, with roughly 4,500 students, was less that 1 percent black and Asian, nearly 2 percent Latino and 98 percent white. Now 20 years later, there are only 2,300 students, of whom roughly 56 percent are black, 22 percent Latino, 18 percent white and the balance Asian.

Between 1970 and 1990, the population of the township shrank nearly 12 percent, to 152,000. Pockets of middle-class comfort became increasingly harder to find. Factories and companies that had once prospered there shut their doors. Looming over these unhappy developments was rampant white flight: as the black population more than impled, the white population plummeted. Whites are still the predominant radial group in Proviso Township, but in the past two decades their numbers have fallen 40 percent.²

In writing the article, Bissinger was allowed full access to the school, classes, faculty, and students. Particularly poignant were statements attributed to students and faculty about race relations at the school. The article depicts a school divided along racial lines, with negative stereotypes pervasive among students and faculty. The racial attitudes and feelings of six white students in an honors course are reported as descriptive of white student sentiment.

"It's like going to hell here," says one. "I get pushed, and because I'm white I can't do anything because there's too many of 'em. I'll get my butt kicked."

"They think they can touch you, they think they can do anything to you."

"I cannot stand the [black] race. I'll never date anyone who isn't white."

"They're always saying 'slave' this and 'slave' that. Sorry, I don't know a slave. I never owned a slave. I'm sick of them throwing it in our face."

When asked to describe a typical black, one of the students responds this way: "Ignorant, rude, loud."

"Ignorant and scum, a lot of poverty, self-righteous, you owe me that, you owe me this, gimme, gimme, gimme," says another.

One student brings up a scene in "2001: A Space Odyssey" when a group of apes goes out of control: "In a typical day in the hallways, that's what it looks like here." Others laugh and nod their heads approvingly. They think it's a good description. They like it.⁴

In a similar manner, the author uses statements from the black students in the same honors class to capture the racial attitudes of black students.

When the [black studenta] talk about whites, it isn't with bitterness but with frustration, almost amusement. Of the seven only one says that the experience of being at Proviso West has made her feel more prejudices against whites. They laugh about the way whites insist on talking to them in what they think is "black dialect"—the way they say, "What's up" and "Hey girl." They also laugh at the way whites, when they go on field trips, stare out the window of the bus and exclaim over graffiti, as if they had never seen it before. The whites strike them as being insulated and utterly unaware of the larger world outside their communities.

These black students also talk about the way other black students often treat them, about how they are accused of being "nerds" and "sellouts" because they are on the honors track. "Why are you in a class with all the

² Ibid., p. 28

Eric L. Eversley, superintendent, Proviso Township High School District 209, Maywood, IL, interview, Sept. 30, 1994, Midwestern Regional Office files.

⁴ Times article, p. 52.

white kids?" they are asked. "Why are you using a white man's book?" . . . "They are really shocked that a black male can read," he says. "They expect me to take the book and throw it in my locker." He described the following interplay between teacher and student as being typical:

"Excuse me, I'm just trying to wake you up so you can pay attention."

"Well, I'm going back to sleep because you ain't sayin' nothing."

Because of the expectations that many white (and black) students have of them, these seven [black] students exist almost entirely in their own world. "We're not accepted by the white people because they think we're not smart enough.... We're not accepted by black people because they think we're too smart. So we just hang with each other."

Bissinger buttresses these thoughts with comments from faculty members. Six teachers are quoted in the article. Only two of the six speak positively of the school and its students. Ann Rebello, a social studies teacher, is described in the article as finding her students "open [and] yearning for attention." Alexis Wallace is portrayed as a teacher "convinced that learning isn't a question of aptitude but of motivation and self-confidence," and is depicted in one particular classroom setting prodding and pushing students to higher achievement.

But the other four teachers have negative attitudes and feelings attributed to them, both about the school and its race relations. Two are alleged to talk of the lack of spirit in the building and the futility of teaching at the school. To one the author alleges an open racial hostility toward minorities. For the other, the author travels back in time 20 years and refers to recollections of the "ominous tone of the meeting" when new racial boundaries for the school were announced.

Mary Frances Berry, Chairperson of the U.S. Commission on Civil Rights, read the article and brought it to the attention of the full Commission at its June 3, 1994, meeting. Berry related:

There was a New York Times Magazine article on May 29, 1994, it was the cover story of the New York Times... As d there is a teacher quoted in the article, who now says that he was taken out of context.... "Some days I come in [and] I just don't want to see anyone black. I have just had it."... And it goes on to talk about his conduct with black students... and his general problems with having black students to teach.

Since I am aware that one of the major problems in education is that students learn when they have teachers who believe they can learn . . . if they have teachers who don't believe that the particular child they are teaching can learn, then the child is unlikely to learn anything. So I was very concerned, not about the article itself . . . but in particular if such a leacher was teaching children.

I didn't understand how they could learn anything from that teacher. 10

The Commissioners discussed the article in terms of national education issues and concernabout a general tendency of some individuals in

⁵ Ibid., p. 53

⁶ Ibid., p 30

⁷ Ibid., p. 31

The author alleges that Bill Paterson, a social studies instructor, "tells the story of a former teacher, now retired, who moved to a small coastal island so he would never have to see another black face. I wish I could leave this year,' says Patterson. There are days I come here when I can hardly face it. I have to force myself." Times article, p. 31. Paterson denies making the comments attributed to him in the article. Students and faculty who remarked on Paterson's alleged comments all denied ever hearing him make such comments or express any negative sentiments about minority students. See comments in parts 3, 4, and 5 of this report.

⁹ Times article, p. 31.

¹⁰ Meeting of the U.S. Commission on Civil Rights, Washington, DC, June 3, 1994, pp. 84–85.

educational institutions to express and hold negative racial and ethnic stereotypes. Berry proposed the Commission send letters to the Attorney General and Secretary of Education to investigate the district and determine if there were civil rights violations. 11

In the discussion that followed it was suggested that in addition to such letters, the Illinois Advisory Committee be asked to monitor the situation at the high school. One Commissioner offered that it was probably the case that this type of situation, if accurate, "was rist limited to one place in Chicago" and expressed an interest in broadening the issue. 12 Another Commissioner suggested the Commission first request the Illinois Advisory Committee examine the matter and offer some recommendations to the Commission. 13

Following the Commission meeting, the Illinois Advisory Committee decided, by a vote of 12 yes and 1 no, to hold a community forum on race relations and equal education opportunity at Proviso West High School for the purpose of obtaining representative teacher, student, and parent perspectives concerning equal education opportunities and race relations at the school. The scope of the initiative was limited to a solicitation of opinion, without analysis of those opinions or an attempt to verify the factual content of either participant statements or the Times article.

Public meetings were held at Proviso West High School on the afternoon and evening of November 30, 1994, and on the afternoon of December 1, 1994. Students, faculty, administrators, parents, and individuals from the community spoke to the Advisory Committee. To avoid selection bias, student and faculty participants were randomly selected. All identified students and faculty cited in the *Times* article were invited to speak. In addition, a public session was held, at which anyone not specifically invited could testify. Comments from every individual who addressed the Advisory Committee are in this document.

Proviso West High School and Proviso Township High School District 209 cooperated completely with the Advisory Committee in this work. The school furnished meeting space to the Advisory Committee, allowed Advisory Committee members to tour the grounds and visit classes, provided school data, and coordinated and facilitated student and faculty participation.¹⁴

Part two of this report gives background and demographic information on the school, students, teachers, and community Part three presents the comments from student participants; the comments are collected into two groups: students selected randomly and student leaders. Part four contains testimony from faculty and administrators; extensive coverage is given in this part to the remarks of the Proviso Township High School District superintendent Eric L. Eversley. 15

Part five presents the statements of parents and individuals from the community. In part six are the Advisory Committee's observations and conclusions regarding race relations and equal educational opportunities at Proviso West High School.

¹¹ The letters, dated June 6, 1994, from Mary Frances Berry, Chairperson, USCCR, to the Attorney General and the Secretary of Education are in app. 2

¹² Ibid, p. 87, Commissioner Russell G. Redenbaugh

¹³ Ibid., pp. 90-95, Commissioner Cruz Reynoso

¹⁴ All students speaking to the Committee had permission from parents and/or guardians to participate at the meeting. The school provided logistical and administrative support to the Committee in contacting students, obtaining parental/guardian permission, and getting student participants to the meeting at the arranged times.

¹⁵ Pursuant to U.S. Commission on Civil Rights administrative procedure, a draft of the report was sent to Eric L. Eversley, superintendent, Proviso Township High School District 209, for review and comment.

Part 2

Proviso West High School

Proviso West High School is in Illinois high school district 200. The Proviso high school township covers 6 square miles. The district has two high schools, Proviso East, located in Maywood, and Proviso West, located in Hillside. Each school has a multicultural population of approximately 2,300 students. The near western Chicago suburban communities of Bellwood, Berkeley, Hillside, Northlake, Stone Park, Westchester, and part of Melrose Park feed Proviso West High School.

The township high schools offer a complete college preparatory curriculum with advanced placement classes in the sciences and mathematics. Junior and senior honors interdisciplinary classes in English and social studies are available. Pre-Scholastic Aptitude Test (PSAT) and American College Test (ACT) preparation classes are held in the district each year. At Proviso West, project LIFT (Learning through the Integration of Familiar Themes) provides freshmen the opportunity to study academics as a series of interrelationships.

Funding for education in the State of Illinois has declined over the last two decades relative to

the rest of the Nation. In the mid-1970s, Illinois ranked 28th in the Nation in providing State funds for education. At the end of the 1992–93 school year, Illinois ranked in the bottom five of the 50 States in providing money for public education. In addition, the State's share of education has decreased. Twenty years ago, Illinois paid 39.2 percent of the cost of education; in the early 1990s, the State paid 33.5 percent of the bill. 1

In addition to funding concerns, Proviso West High School receives a freshman student body from the feeder elementary schools in which more than half of the students read beliggrade level. Results of the Stanford Diagnostic sading Test administered in September 1984 to embers of the Proviso West High School class of 1996 show 310 of the 575 (54 percent) ninth graders tested scored at grade 8 or below.2

Student Background, Enrollment, and Achievement

Racial/Ethnic Background and Enrollment

The racial and ethnic composition of the student body has changed over the last two decades. In 1970 the student population was almost

Rayla Attice Baska, ind Rate Left for Art	

	V./hite	African Amer.	Hispanic	Asian Amer.	Notive Amer.	Total
Proviso West	18.4	56.4	22.2	3.0	0.0	2,303
District 209	11.8	69 .3	16.5	2.3	0.0	4,427
Illinois	65.2	20.7	11.0	2.9	0.1	

Source: Proviso Township High School District 209, 1994

Proviso Township High Echool District 209, "Your High Schools Report," Winter/Spring 1994, p. 1.

Dale Crawford, memorandum to the superint andent, Dec. 16, 1904, Midwestern Regional Office, USCCR files.

TABLE 3
Limited English Proficient and Low Income
Student Rates

•

	Limited English	Low income
Proviso West	6.6	17.1
District 209	3.5	22.7
Illinois	5.2	33.5

Source. Proviso Township High School District, 1994. Report.

exclusively white. In 1980 the student population was 3,252 and still predominately white: 2506 whites (77 percent), 457 African Americans (14 percent), 220 Hispanics (7 percent), and 69 Asians and American Indians (2 percent) Today, the total student enrollment has dropped to 2,303, and white students are a numerical minority.

The enrollment at Proviso West for academic year 1993-94 was 2,303. African Americans are the majority of students, 56.4 percent of the total student population. Hispanics are the second largest group, 22.3 percent; whites are 18.4 percent of the student body; the Asian student population is 3 percent.⁴

Proviso West has a higher enrollment rate of students who are limited-English proficient than the State rate. The low-income rate at the school is below the State rate. Limited-English proficient students are those who have been found to be eligible for bilingual education. Low-income students are from families receiving public aid, living in institutions for neglected or delinquent children, being supported in foster homes with public funds, or eligible to receive free or reduced-price lunches.⁵

Attendance

The school's dropout rate is higher than that found in the rest of the district and in Illinois. The dropout rate is based on the number of students in grades 9-12 who dropped out during the 1993-94 academic school year. The school's attendance rate, however, is higher than in the district, but lower than statewide. A perfect attendance rate (100 percent) means that all students attended school every day. No chronic truancy, defined as students absent from school without valid cause for 10 percent or more of the last 180 school days, was reported at Proviso West high school during the 1993-94 school year.⁶

TABLE 4
Attendance Rates, Dropout Rates, and Chronic Truancy

	Attendance	Dropouts	Chronic truency
Proviso West	91.7	9.3	0
District 209	9 0.7	7.7	20
Illinois	93.2	7.0	MA

Source: Proviso Township High School District 209, 1994

Report.

³ Proviso Township High School District 209

⁴ Proviso Township High School District 209, 1994 School Report Card, p. 2 (hereafter referred to as 1994 Report). Data for the 1993-94 academic year data is used to comport with school data at time of the Times article.

^{5 1994} Report, p. 2.

⁶ Ibid., p. 2.

TABLE DE CARREST OF THE SECOND

76.6
71.1
78.0

Snurge: Proviso Township High School District, 1994. Report.

Graduation Rate

The graduation rate at Proviso West High School for the 1993–94 academic year (76.6 percent) is essentially equivalent to the State graduation rate (78.0 percent). The graduation rate compares the number of students who enrolled in ninth grade in the fall of 1990 with the number from that group who actually graduated in 1994. Adjustments to the rate are made for students who transferred in and out of the school. The graduation rate does not include students who graduated in the summer of 1994 or those who took more than 4 years to graduate.

Standardized Achievement Testing

The Illinois Goal Assessment Program (IGAP) test is administered to all students in the State during their sophomore year in reading, mathematics, and writing, and all students in their junior year in science and social science. Average IGAP scores in reading, mathematics, science, and social science are reported on 0-500 scale; scores for writing are on a 6-32 scale.

The percentage of those tested at Proviso West was generally higher than the rate of students tested statewide. Sophomores at Proviso West scored at the State average in writing and below the State average in reading and mathematics. Juniors at Proviso West scored lower than the State average in science and social science.

College Teeting

Prospective college students at Proviso West take the ACT. ACT scores range from 1 (lowest) to 36 (highest). Proviso West students scored lower when compared to all students in Illinois taking the ACT for the academic year 1993-94. The composite score for Proviso West students is 18.1,

TABLE 6

IGAP Average 5 condition Relation, Mathematical Westing 5 conditions of 50 cones

	Reading		Mathematics		Writing	
	Score	% Tooled	Score	% Tooted	Score	% Tested
Proviso West	210	80 .7	201	82.2	25.3	79.1
District 209	186	81.7	190	80.4	24.1	74.8
Illinois	244	81.9	254	81.3	25.3	78.9

	Sci	ience	Socia	al Science
	Score	% Tested	Score	% Tested
Proviso West	210	86 .2	199	87.4
District 209	201	76.8	185	79.8
Illinois	256	82.9	245	83.6

Source: Provise Township High School District 209, 1994 Report

⁷ Ibid., p. 8.

s Ibid

	ACT —All students				
	Score	Band	Test taker %		
Proviso West	18.1	17.6-18.6	66.6		
District 209	18.0	17.6-18.4	63 .0		
Illinois	21.0	•	62.4		

ACT - Students who completed core program

	Score	Band	Test tales %
Proviso West	19.2	18.4-20.0	28.3
District 209	19.4	18.7-20.1	23.8
Minois	22.9	•	27.5

^{*} State score bands are not shown because they are very narrow

Source: Provise Township High School District 209, 1994 Report.

with a score band of 17.6–18.6; the composite ACT score for all students in the State was 21.0.

Score bands can be used to compare composite scores. If the score for the State falls within the score band for a school, then there is no significant difference between the school score and the State score. However, if the State score is outside the score band for the school, then there is significant difference between the school score and the State score.

Students at Proviso West who had completed a core high school program also scored lower on the ACT than similar students in the State. A core program is a high school program that includes at least 4 years of English and at least 3 years of mathematics, social studies, and natural sciences. Generally, students who complete core programs earn higher average scores than those who had less than core programs. Students at Proviso West who completed a high school core program had a composite score of 19.2 compared to a composite score of 22.9 for students in the State who completed a core program. A higher percentage of

students at Proviso West took the ACT (66.6 percent) than the statewide average (62.4 percent). The rate of test taking for students at Proviso West who completed a core high school program was 26.3 percent, compared to 27.5 percent of similar students in the State.⁹

Teacher Background, Tenure, and Class Size

Recial/Ethnic Diversity

At the start of the 1993-94 academic year there were 144 teachers at Proviso West. Teachers include all school personnel whose primary responsibility is listed as that of classroom teacher on the State Teacher Service Record file. The race/ethnicity of the faculty is: 125 white (86.8 percent), 16 African American (11.1 percent), 2 Hispanic (1.4 percent), and 1 Native American (0.7 percent).

Recent hiring rates have a similar race and ethnic composition. For the 1993-94 academic year, 19 teachers were hired including: 16 whites (85 percent), 2 African Americans (10 percent),

⁹ Ibid., p. 7

¹⁰ Ibid., p. 3.

TAPET STEEL COSTS But Kan gind

	Milita	African Amer.	Hispanic	Asian Amer.	Native Amer.	Total
Proviso West			1.4	0.0	0.7	144
District 209	86.1	12.0	1.1	0.8	0.0	266
Illinois	84.5	12.5	2.4	0.6	0.1	NA

Source: Provise Township High School District 209, 1994

Report.

TABLE 9 Mean Leadard Appliance

Proviso West 15.7 years
District 209 17.2 years
HS Districts 17.5 years
Illinois 15.4 years

Source: Proviso Township High School District 209,

1994 Report.

and 1 Hispanic (5 percent). For the 1994-95 school year, 27 teachers were hired including: 23 whites (85 percent), 2 African Americans (7.5 percent), and 2 Hispanics (7.5 percent).

These hiring rates, though lower than the race and ethnicity rates of the student population, exceed the hiring rates of minority secondary teachers in Illinois. In the period July-August 1993, the race and ethnic ratios of new secondary teacher hires were: 96.1 percent white, 2.7 percent African American, 0.9 percent Hispanic, and 0.3 percent Asian.¹¹

Salary

The average teacher salary in high school district 209 for the 1993-94 academic year was \$53,788, essentially equivalent to the average salary for all high school districts in Illinois, \$64,068. The district average salary is slightly lower in comparison to other large-size high school dis-

tricts in the State, which is \$59,490. The average teacher salary in Illinois for the same period for all teachers was \$39,545. Teacher salaries include various monetary benefits and compensation such as tax-sheltered annuities, retirement benefits, and extracurricular duty pay. 13

TABLE 10 Teacher Salary and Per Papel Cord

	Teacher salary	Cost per pupil
	1 993-84	1992-83
District 209	\$53,788	\$8,786
Type*	\$54,068	\$8,518
Size * *	\$59,490	\$9,592
Illinois	\$39,545	\$5,579

^{*} Average for all high school districts in illinois.

Class Size

Average class size at Proviso West High School (21.8) was slightly higher than the average high school class size in Illinois (19.9) for the 1993-94 academic year. Average class size is the total enrollment for a grade divided by the number of crasses for that grade reported for the first school

^{**} Average for all large high school districts in Illinois. Source: Provise Township High School District 209, 1994 Report.

¹¹ Illinois Department of Education, department of teacher certification.

¹² Ibid., p. 4.



Proviso West 21.6 District 209 21.4 Illinois 19.9

Source: Proviso Township High School District 209, 1994 Report.

day in May. For high schools, the average class size is computed for the whole school, based on average class sizes for the second and fifth periods.¹³

School Discipline

African American students have the largest suspension rate in the school district. Multiple suspensions were given to 1,150 students in the district during the 1993–94 academic year: 894 African American students (77.7 percent), 174 Hispanic students (15.0 percent), 73 white students (6.3 percent), and 9 Asian students (1 percent). Eight hundred sixty-two (862) students in the district were suspended once and only once: 636 African American students (73.8 percent), 141 Hispanic students (16.3 percent), 77 white students (8.9 percent), and 3 Asian students (1 percent).

Suspension rates at Proviso West High School are similar to the district figures. Multiple suspensions were given to 751 students at Proviso West High School during the 1993-94 academic

year: 536 African American students (71.4 percent), 143 Hispanic students (19.0 percent), 66 white students (8.8 percent), and 6 Asian students (0.8 percent). Four hundred and seventy-four (474) students at Proviso West were suspended once and only once: 290 African American students (61.2 percent), 113 Hispanic students (23.8 percent), 64 white students (13.5 percent), and 7 Asian students (1.5 percent).

The racial and ethnic breakdown of the total 1,225 suspensions during the 1963-94 academic year is: 826 African Americans (67.4 percent), 256 Hispanics (20.9 percent), 130 whites (10.6 percent), and 13 Asians (1.1 percent).

TABLE	. 1			,
S. Hill		• B	of their	et 🐫 📜

Students suspended ence and only once

	Number	Group rate
African American	290	61.2%
White	64	13.5
Hispanic	113	23.8
Asian	7	1.5

Students suspended more than once

	Number	Group rate
African American	536	71.4%
White	66	8.8%
Hispanic	143	19.0%
Asian	6	0.8%

Source: Midwestern Regional Office, USCCR, from Minois State Board of Education and Proviso West High School deta.

¹³ Ibid., p. 8.

Part 3

Student Comments

Porty-five students made statements about race relations and equal education opportunity at Proviso West High School. The student selection process was designed to produce an accurate representation of student opinion concerning the school's racial climate and educational opportunities. Two groups of students were selected: a random sample of 30 students and a random selection of 12 student leaders. Three other students volunteered to speak to the Advisory Committee.

The random sample of student participants was generated from a list of student identification numbers active at the start of the 1994–95 school year. Forty-five student numbers were selected, the last 15 selected for use as alternates. Nine of the alternates were used, substituting for 6 students who had left Proviso West and 3 students who could not be interviewed. School policy required all selected student participants to have a signed permission slip from a parent or guardian before speaking to the Advisory Committee. Nineteen of the students in the random sample did not provide parental/guardian permission at the time of the meeting, and were interviewed at a later date by Commission staff.²

The random sample closely resembled the race and ethnicity of the student population. The racial/ethnic composition of the student population is: 56.4 percent African American, 22.2 per-

cent Hispanic, 18.4 percent white, and 3.0 percent Asian. The racial/ethnic composition of the random sample is: 18 African American (60 percent), 6 Hispanic (20.0 percent), 5 white (16.7 percent), and 1 Asian (3.3 percent).

The statements of the students are presented without comment, letting the students' words speak for themselves. The testimony of some students before the Advisory Committee is condensed, while the remarks of others, due to their brevity, are presented in entirety.

Selected Student Leaders

George Acevedo, class of 1995:

I would just like to say that when talking about racism I feel that everyone has a certain degree of racism, but when it comes to society, whether it be in school or in community, I feel that it can be greatly exaggerated. From the New York Times [Magazine] article that came out in May, I felt that that was greatly exaggerated and I just wanted to point that out. I think we are more proud [of our school] because we have ethnic diversity.

Brandi Armstrong, class of 1995:

What I have experienced at this school is called prejudice not racism.... I personally have experienced racial prejudice towards me, but you have to look at things like this [with the understanding] that somebody is

Opportunity was provided to all students to participate, either by making statements directly to the Committee during public sessions or by providing written comments to the Committee through the Midwestern Regional Office, USCCR.

² The 19 students ultimately provided parental/guardian permission and were interviewed by staff from the Midwestern Regional Office, USCCR. Records of the interviews are in the files of the Midwestern Regional Office, USCCR, Chicago, IL. Two students did not want their statements made public.

³ Student comments presented in their entirety are denoted by ***.

Testimony before the Illinois Advisory Committee to the U.S. Commission on Civil Rights, community forum, Nov. 80 and Dec. 1, 1994, Proviso West High School, Hillside, IL, p. 71 (hereafter referred to as Transcript). George Acevedo participates in varsity school athletics, other student organizations, and is a member of the National Honor Society.

TABLE 7.5 Student leaster Participants

Activity George Acevedo **Athletics** Brandi Armstrong **Athletics** Student council Clarise Bennett Class president Kizzy Elem Student council Hilda Farah Class president Zachary Hudson Student council Manui McCullough **Athletics** Kelly Morgan Class secretary Carolyn Moss Jesus Reves **Athletics** Heidi Schulz **Athletics** Student council Lorie Sojak

always going to dislike you for various reasons, you cannot stop that....

The problem that I perceive is prejudice, not only at my school but also at times in my community. Of myself, I have bumped into someone and I am short and black. And if I bump into someone that is of an Hispanic or maybe of a Caucasian or someone of an Asian [ethnicity], they might automatically think I'll get an attitude . . . and I am not like that at all. That's just a stereotype.⁵

Clarise Bennett, class of 1996:

I have never encountered any recial problems at Proviso West, not during my 3 years here. As for other groups, I don't have a problem with any recial group. . . . I don't think there is a magnitude to the [race] problem because I think everything is blown out of proportion. There are race problems, but no more than there are in any other schools in any district. . . .

[The Times article blew the race problem here] out of proportion. It was wrong . . . that everything that was good or fair about the school was not mentioned. They only mentioned stuff that was negative and the few things that are negative, [the author] made into a huge deal. There was no positive, just negative.

Kizzy Elem, class of 1996:

I think there will always be racism, but I don't think there is much racism at [this] school. In the past there have been fights with different races. And some people say things. Like last year we had "riggers" [written] on the walls and on the lockers. . . . And in the classrooms people talk about each other, but it is not that much. . . .

In the [lunchroom] there was a time where the white students would be on one side, the black people would be on the other side. They mixed together sometimes.... Sometimes they know each other.

Hilda Farah, class of 1995:

At our school in my eyes I don't see any race relations [problems]. I mean sure we have majority black and Hispanic, [but] in my classes, in the halls, and in the lunchrooms I don't see any racial discrimination....

[The article] was blown way out of proportion. It did not respect what our school represents at all. When we first heard of this reporter coming into our school, we thought he was going to make it a real positive issue. He even said so, and he went and just back faced us. He took parts out of quotes and made them look negative which I don't think was right... I was in that American studies class which he studied very carefully. And our whole class was mad... about the remarks that were made in the [article].

Zachary Hudson, class of 1997:

For our school I don't see any prejudice at all. I get along with everybody and everybody tends to like me and I tend to do what I have to do.... I feel like if you

⁵ Ibid., pp. 74 and 77. Brandi Armstrong is an honor student, a member of the school's track, backetball, and volleyball teams, and also a student tutor.

⁶ Ibid., pp. 17 and 19. Clarise Bennett is president of the student council and involved in other school activities.

⁷ Ibid., pp. 47-48 and 51-52. Kizzy Elem is president of her class and a member of the school's varuity basketball team.

⁸ Ibid., pp. 32 and 34. Hilda Farah is on the executive board of the student council.

come here, come respectful and [act] in a respectable manner and do what you are supposed to do, people will respect you. I never had any trouble with anyone and I don't see any kind of prejudice here because I am very respectful and I give respect and to a certain extent I get respect. . . .

Everybody gets along as far as I see. For what I see we come here, have a good time. . . . You are going to run into some negativity, [and some] people try to get you in trouble or something like that, [but] you have that everywhere. 9

Manui Ann McCullough, class of 1995:

There are problems [here]. I know there are definitely some problems between other people here...but I don't have any problems with anybody. I get along with almost everybody....

As student council members, our job is to try and get [problems] stopped and get the students involved and get everyone together. And we are working really hard this year and we have done a lot of things to get school involvement in school spirit and people together. We worked a lot harder this year trying to make things stop, but it has gotten gradually worse as the years go on. . . .

It is a totally different atmosphere here than at another school. But it is also more of an atmosphere in which you get to relate to more people and more cultures and everything. So you get to know other people and get to understand them better. ¹⁰

Kelly Morgan, class of 1996:

I myself feel that there are no race relation [problems] at this school. I feel that we are all here for one purpose, to learn, and any problems that arise from that aspect is mainly out of ignorance and from people that aren't

here to learn, they just don't want to deal with the aspect of education. So there are really no race relation [problems]; everything is of a one-on-one contact with people....

I would say a majority of [the student body] would agree with what I've just said just because most of the people here are open minded about their community and this school, and we do take pride in Proviso West. 11

Carolyn Moss, class of 1995:

My views on race relations [at Proviso is] there are a variety of races at Proviso and there is ... racism—not a lot of racism, but there is [some] in the school as in other high schools and I feel it cannot be [eliminated]. . . . Not everyone agrees to the same things and certain . . . minds aren't as open as other people's are. . . .

Ewing: What specific practices would you like to see removed from the school grounds?

Moss: The fighting. The fighting I would like to be gone. The negative images of how much racism there is in this school [comes from this and] there is not that much. People or students from other high schools tend to look down at Proviso West because of that and they don't know the [entire] story, the true stories... behind the stories that are in the newspapers. 12

Jesus Reyes, class of 1996:

My opinion is that if you are walking down the hall and you bump into somebody, you might get a race comment or something. But like if you are in sports [there is no problem]. I'm in wrestling and soccer and wrestling is mostly black and I get along fine.... I get along with black people, white people, Hispanics, Asians.... But I'm a junior [and] my freshman and sophomore year I did hear a couple of racial comments. 13

⁹ Tbid., pp. 33 and 38. Zachary Hudson is president of the class of 1997 and on the executive board of the student council.

¹⁰ Ibid., pp. 41, 45, 46. Manui McCullough is a senior and secretary of the student council.

¹¹ Ibid., p. 79. Kelly Morgan is a member of the school band, newspaper, key club, and the swimming and soccer teams.

¹² Ibid., pp. 62 and 65. Carolyn Moss is secretary of the senior class.

¹³ Ibid., p. 29.

Heidi Schulz, class of 1995:

There are race problems (here), yes, and no, there aren't problems. Each group of friends is different. There are black friends who like white people, white people who like black people. Like for me, I was on the track team. I was the only white person sometimes or one of the few. We got along great, absolutely nothing at all. Then you go in the hallways and sometimer there are problems. But then again there are others who don't give you problems, who don't say racial clurs if somebody knocks you in the hallway. The [race] problems exist, but then they don't as well.

Yandle: Are the issues addressed in [The New York Times Magazine] article or the things reported in that article . . . overexaggerated or is there any factual basis?

Schulz: There is some truth. I would say 75 percent of it is blown out of proportion, but yes, problems do exist and we can't, I can't, say that they do not and that this is a great place.¹⁴

Lorie Sojak, class of 1995:

I don't have any problems with the race relations here. I mean there are people who are ignorant, but that is any race.... I have a lot of black friends, white friends, Hispanic friends, whatever....

I think Proviso is a good school. I don't consider race relations a problem [here]. 15

Students Selected by Random Sample

Bernada Baker,* class of 1998:

I have never had any racial situations. It's fine here. I don't have any problems. It's a nice school and I am doing better here than at my previous school because my teachers are enforcing the work. 16

Lillian Barbari,* class of 1998:

I think everything is fine here. No one is racist here. Maybe because the majority of the school is black, people who do not go here think it is a problem. But everything is okay.¹⁷

Katia Becerra,* class of 1997:

Everyone gets along here. Once in awhile there are fights, but those are personality issues, and it's African Americans against African Americans and Hispanics against Hispanics, for instance. The teachers I know are not racist. I have heard rumors of racist teachers but I have not had an experience with prejudice from a teacher. 18

William Brown,* class of 1998:

It's fine with me. The school is fine. The teachers are nice. Everything is okay. 19

Shane Clarke,* class of 1996:

I feel students are more racial here than the faculty. The kids move in segregated groups. Me and my friends are sometimes called "white boys" by the black students particularly when we sit at our own table at lunch. I don't think it is more than that. I feel there is a problem of racism here and it should be dealt with. I feel like a minority here. he school is a great place to get an education but it should be more united. If we got rid of

¹⁴ Ibid., pp. 25 and 30.

¹⁵ Ibid., pp. 41 and 46. Lorie Sojak is a senior and treasurer of the student council.

¹⁶ Interview, Dec. 15, 1995, Proviso Wes. High School, Hillside, IL, Midwestern Regional Office, USCCR files (hereafter referred to as Interview).

¹⁷ Ibid.

¹⁸ Ibid.

¹⁹ Ibid.

the racism there would be more spirit. Like at dances, the whites don't attend because of the music. ²⁰

Henry Alexander Day, class of 1997:

Some people think (there are racial problems at Proviso West) and some people think there aren't. In my opinion I really think it is not; because it is just in the minds of certain people that there may be racial [problems] in this school, but to me it is really not.²¹

Laura Deyo,* class of 1998:

I think it is okay here. It's all right with all the races. There is each race here, black, white, Mexican. It's fun to get to know all of them. I like to make friends with all the people here.²²

Sandra Franco, class of 1995:

I think, for me, I have never had a racial problem. . . . I don't think there is a lot of racial problems here. . . .

Roberts: Did you have any fears of coming here?

Franco: No, [at first] I thought I was going to be the only Hispanic. But I did not . . . hear anything negative. . . .

Mathewson: Could you compare your earlier experience at [West Lyden] with your experience at Proviso West?

Franco: I like it better here because I have more opportunities. . . .

Schwartzberg: In the cafeteria do students tend to sit separately?

Franco: Sometimes, [but] nothing happens [if that is broken] because we just move to another table or we sit there at the same table.²³

Kevin Gill.* class of 1996.

Name	Rece/ethnicity
Bernada Baker®	African American
Lillian Barbari°	Hispanic
Katia Becerra®	Hispanic
William Brown ^o	African American
Shane Clark®	White
Henry Alexander Day	African American
Laura Deyo°	White
Sandra Franco	Hispanic
Kevin Gill ^o	African American
Tanesha Griggs °	African American
Ayanna Hannaho	African American
Nancy Kernan	White
Jason Knight ^o	African American
Jeffrey Langeland ^o	White
John Marszaleck	White
Garrett Matthews ^o	African American
Octavia McCadd ^o	African American
Chevon Nightengale	African American
Maribel Pinedo	Hispanic
Severin Richardson	African American
Anu Saini	Asian American
Terrence Spencer	African American
Jason Stewart®	African American
Rekida Thomasº	African American
Myra Tidwell	African American
Sugar Wright	African American
Marco Vargasº	Hispanic
Esther Vega®	Hispanic

Students in this table are those selected by random sample. Two students making statements did not want to be identified or have their statements made public.

* Students giving statements subsequent to the community forum.

²⁰ Ibid.

²¹ Transcript, p. 308.

²² Ibid.

²³ Transcript, pp. 313, 314, 315, and 316.

Well, to me, I think that some people in this school are prejudiced. They don't show it to me, but I see it, it's there. People, if you ask them if they are prejudiced, will deny it, but I can see it's there by the way they act. I never knew it was here at first, but last year I ran into it for the first time. The teachers are okay, though, no problems there.²⁴

Tanesha Griggs, class of 1998:

I don't know about any racism here.²⁵

Ayana Hannah,* class of 1996:

I think everything is okay because it is normal here. I feel I get along with a lot of different races. I went to school previously where there were all hinds of people. Here the people I see in the hallways get along; there is no race problem. It is a nice school and teachers are good. I have never had a situation where teachers treated me or anyone else unfairly because of color. 25

Nancy Kernan, class of 1998:

I do not think there is a racial problem [here]. I have never heard any comments from any teachers or any students relating to racial problems. There is none. And I don't think the newspaper article did justice. It just wrote bad things about West which were untrue....

It is mostly people that don't go here that say things about [us] because they don't know what really goes on here. They think that it's a bad situation because it's [got] a lot of black people and Mexicans. But it's not bad at all.²⁷

Jason Knight,* class of 1997:

The race situation is better here than at other schools I have been to. This is my first year here after being at another high school. The people get along better here. There is not much tension here. It is just different here.

You do get along easier with your own race, and by my going to Proviso West, where it is mostly African American, it's easier for me than at the other school I went to which was mostly white.²⁸

Jeffrey Langeland,* class of 1996:

I don't think the problems here are racial. It's like any other school, the school's problems are not racist, but gang problems. Overall the school is okay.²⁹

John Marszaleck, class of 1997:

I am a sophomore here and I really don't think there is any racial problem here. This is my second year and it seems pretty fine. I like the school and everything is fine. . . . I'm in plays here at the school and work with blacks, whites, and everybody. And we just come together where you don't think about race. 30

Garrett Matthews,* class of 1995:

I have no racism problem here with any of my teachers. I'm not saying (though) that there aren't any racism problems at Proviso West High School. 31

Octavia McCadd, class of 1997:

I have never seen a lot of racism at the school. I haven't had any racism shown toward myself personally. The school is all right. I am getting my education and see others getting their education. Teachers try their best and will give help to students if they want it. If there is anything racist, teachers will discipline the student or

²⁴ Interview, Dec. 15, 1994.

²⁵ Ibad

²⁶ Ibid.

²⁷ Transcript, pp. 289 and 292.

²⁸ Interview, Dec. 15, 1994.

²⁹ Ibad.

³⁰ Transcript, pp. 290 and 291.

³¹ Written comment in response to verification of original interview statement made on Dec. 15, 1994.

show them it wasn't the way to act. The teachers do not show any racism at this school.³²

Chevon Nightengale, class of 1995:

I feel that people at this school keep to themselves, like different races stay in different groups. Different groups like Puerto Ricans and Hispanics and whoever, blacks or whatever, stay in separate groups. So it is not really like bonding together.

For myself I feel that I make friends with all races and stuff because that is the type of person I am [but]...I witness [this when I] walk through the halls and notice that different groups are in different groups with the same kinds of people. I guess it is because if you are in the same cultural group, you have the same things in common with the other people. So they just stick together. I feel they can't open their minds to other cultures.

I understand that some of the black students, upperclassmen, some of them,... are disruptive. But they are not given a chance as ones that want to learn something. Mostly they pay more attention to the white kids.... That's what I see.

Yandle: Have you experienced or observed things you felt were motivated by racial prejudice or racism?

Nightengale: Well, yes. When I was in my freshman year I was on the softball team, and for some reason I don't feel that, I'm not going to say any names, but the coach of the softball team mostly played the white students that were on the team instead of the blacks. It was like one black was playing. I noticed that when I had tried out before for the softball team. That's mostly it.... 33

Maribel Pinedo, class of 1995:

I don't see any discrimination or anything like that [at Proviso].³⁴

Severin Richardson, class of 1995:

In the time I have been here, I haven't seen any racism . . . in this school. . . . I haven't seen any really racial conflict, but at [my previous school] Lakeview, it was a gang problem and it was blacks against Puerto Ricans. 36

Anu Saini, class of 1998:

Lyon: Do you believe that anybody has picked on you because of your race or culture?

Seini: Yes, some boys in my class.

Lyon: What types of things do they say or do?

Saini: They call me names and stuff because of my religion.

Lyon: And when this happens what do you do?

Saini: I just go tell the teacher [and] they get in trouble.²⁶

Terrence Spencer, class of 1995:

This is my first year here and since I have been here I haven't seen nothing too much racial going on. There's more blacks and Hispanics here, but there is really no difference....

Scales: In the group of African Americans . . . is that group mixed in terms of the community?

Spencer: It's not like we just hang out solely with black folks over here, Mexicans over here, whites over there. If we feel like talking to somebody, we converse . . . It's

³² Interview.

³³ Transcript, pp. 301-02, and 304-06.

³⁴ Ibad., p. 313.

³⁵ Thid., pp. 318-20, and 323.

²⁶ Ibid., p. 297.

not like I don't like you because you're Mexican; nothing like that. We talk to everybody.²⁷

Jason Stewart,* class of 1996:

This school does not have much racism to my knowledge. If there is it's the paraprofessional staff. If a black person walks down the hallway, they will ask for their pass. But I know of situations where, for instance, a white female walked down the hallway, no pass in her hand, and she was not asked for a pass. As for my teachers, if they have racism inside, they don't show it. 30

Rekida Thomas,* class of 1998:

I don't think there is racism here. I have only been here for a few months, but I don't hear anybody saying anything racist. The school is fair to me. I like high school and I like this school.⁴⁰

Myra Tidwell, class of 1995:

I don't feel that there is a big racism problem here. I mean there is always going to be people who don't like you for their certain reasons. It may be because of your skin color. There are people here just like there are people at other high schools... that are going to be like that and there is nothing you can do to change their point of view of you. But most of the problems we have at Proviso don't stem from race problems, it stems from one-on-one people having disagreements with each other, not with their race itself and sometimes it gets blown out of proportion. People look at it the wrong way and that is what they see, but it is not like that all the time. 41

Marco Vargas,* class of 1995:

I think there is no racism here. The [New York Times Magazine] article's headline, "We're All Racist Now," is not true for me, and it is not true. 42

Esther Vega,* class of 1995:

I think that there are a variety of cultures at Proviso West High School. Some students try to ignore it, but when the newspaper article came out, blacks were upset, Hispanics were upset, but the whites were not talking. I had Mr. Paterson for class. I know him and the things said about him in the article were taken out of proportion. Some other students, though, who had never had him were influenced about him by the media.

I ran for homecoming queen this year and I felt I would be booed at the assembly because I was Hispanic. As it turned out I was not booed and I won. As I became more aware of student attitudes, I felt more support from lots of different students. Students expressed feelings to me and I got a lot of good feedback.

I never have felt any prejudice here. I have African American and Caucasian friends. I feel individual problems of racism exist at Proviso, but there is not an overall race problem at Proviso West. I have never been exposed to prejudice here. If I was I would not recognize it [as a school problem], but would feel it waz an individual problem. Anybody can achieve whatever they want here at this school. You can't blame it on your culture or color.⁴³

Sugar Wright, class of 1995:

The only problem I have, really I don't have a problem, I mean with green or black, the only problem I have is the school rules. I feel black students make it hard for themselves by doing bed things. [There is a] rule . . . [about] if you miss a certain amount of days. Some students would like to graduate on time and the [rules]

³⁷ Ibid., pp. 318-19 and 321-24.

³⁶ The paraprofessional staff is security and hall monitors.

³⁹ Interview, Dec. 15, 1994.

⁴⁰ Ibid.

⁴¹ Transcript, pp. 62-63.

⁴² Interview, Dec. 15, 1994.

⁴³ Ibid

are making it harder for some students to do that. If you are late to first period you can't get in the class-room; you have to go to the cafeteria. If you miss 8 days, [or] are late 8 days, you won't be able to get a grade. That's the only problem I have.⁴⁴

Volunteer Student Comments

Kim Lasky, class of 1995:

I was interviewed by Mr. Bissinger for the New York Times article and I did say that "We're all racist now." I am not going to deny that I made that comment. But Mr. Bissinger took it out of context and did not include the rest of it which made all the difference in the world. After I said that, I went on to say that I wasn't particularly talking about blacks, I wasn't talking about whites, I wasn't talking about Hispanics. I was saying that everybody takes a part in racism.

There is racism at Proviso, but there is racism everywhere you go. Mr. Bissinger chose to ignore that fact... and made me look like I was saying all white people are racist at Proviso. That was not what I was getting at, at all.... My parents gave me the option of where I wanted to attend school and I chose Proviso. If I were a racist, I would not be here....

Maybe I should have used the word prejudice, because everybody has someone they don't like. Everyone has different opinions about different people. So maybe I should have used the word prejudice, because I know that racism means hating an entire group of people.

[When I said "we're all racist now"] I meant that people tended to separate in their own groups, and in doing so they are—in a way——cist, because like everybody has said so far, that's the way it is at Proviso: the blacks mostly hang out with the blacks, the whites hang out with the whites, Hispanics hang out with the Hispanics. It is either that way or it is the students in the honors classes hang out with the kids in their classes....

[Racism] is not a big problem at [Proviso]. Mr. Bissinger made the whole article out that that was the only problem we have here and that there were no good things at Proviso. When my family and I talked to him, we made sure that he knew that we were the biggest supporters of Proviso . . . and he just chose to ignore all that.⁴⁵

Jeff VanderMolen, class of 1995:

I happen to be editor-in-chief of the paper here at Proviso and the following month after the article came out I had my letter to the editor published in the New York Times Magazine. 46 [I wrote] because it . . . disturbed me to read [the article].

We had a meeting of our [newspaper] staff the month after, actually the next day after it came out, and we all came to the conclusion that none of the people quoted in the article were called back to have their quotations checked, even Mr. Paterson. . . . And I know for a fact that the young lady that was quoted for the headline ["We're all racist now"] was not called back. So she was totally misquoted and her quotes are taken out of context.

That was totally wrong. He turned her words all around. She may have said those words, but not altogether in one complete sentence. He turned it all around and made it seem like she said that.⁴⁷

Jim Redden, class of 1995:

I have not seen too much racism personally. Nobody calls me names and I don't call anyone names. It is pretty cool here. There may be other students for whom this is not true, but for me I have no problems, and there are no racial fights or racial namecalling.

The [Times] article was stretched. There is a race problem here, but there is a race problem everywhere. I don't think Proviso is as bad as he [Biesinger] said it was.⁴⁸

⁴⁴ Тгаластірі, р. 307.

⁴⁵ Transcript, pp. 329, 330, 331, and 334.

⁴⁶ Jeff VanderMolen's lutter is in app. 3.

⁴⁷ Transcript, pp. 228-29, and 231.

⁴⁸ Interview, Dec. 15, 1996.

Part 4

Faculty and Staff Comments

Seventeen faculty and staff gave testimony on the racial climate and equal educational opportunity at Proviso West High School. Fifteen of the presenters were randomly selected from a list of faculty and staff; two others volunteered to speak. Two of the faculty included in the sample had been quoted in the Times Magazine article. Seven of the faculty and staff originally selected declined to speak to the Advisory Committee and were replaced by alternate faculty/staff selections.

The final list of faculty/staff included the departments: business education (1), english (4), guidance (4), library (1) physical education (2), paraprofessional staff (1), social studies (1), and special education (2). One assistant superintendent and one school social worker also testified.

In addition, Eric L. Eversley, Ed.D., superintendent of Proviso Township High School district 209; Leo Banks, president, board of education, Proviso Township High School district; and C. Rebecca Montoya, Ed.D., principal of Proviso West High School made presentations. Opportunity was also extended to the faculty members quoted in the *Times Magazine* article to either speak directly to the Advisory Committee or submit written statements.

Administration

Eric L. Eversley, Ed.D., superintendent:

The publication of the New York Times Magazine article... created a firestorm of reaction and that reaction continues. Most of the reaction centered around several racially offensive comments which were attributed to individuals and generalized by some to be reflective of the attitude or perspectives of all district

employees. That generalization . . . is wrong and it misrepresents our school and our school district.

Whether those racially offensive comments were made or not, I cannot say. That is a matter known to the individuals to whom the comments are attributed, to the author of the article, and to the Almighty. To the extent that those comments reflect the feelings of any individual in our district or to the extent that they would lead to discriminatory behaviors on the part of any individual, they are unacceptable, they are intolerable, and they will not be supported by the school district. I stated that position at the time of the article's publication, and I resterate it here for you today. [At the time] the article appeared . . . I stated a position which is still a position I maintain. . . . I would like to read a portion of that communication to the faculty and the staff of our school district.

"While we may all have our own personal disappointments with the article, I encourage that we not dismiss it. To do so would cause us to miss an opportunity for personal and organizational introspection, organizational development, and personal and professional growth. I trust that it is no surprise to you that I am distressed by images of students not being constructively engaged in instruction. I am concerned about even any suspicion that our work with students is driven by low expectations or negative perspectives about our students as human beings.

"Clearly our school environment, our institutional operations, and the behavior of adults and students must be free from the harmful effect of bias, including racism and sexism. We will never be the school district of excellence if we allow bias to place artificial limits on us. To the extent that bias, prejudice, and discrimination exists, the intellectual and social development of our students as well as the personal and professional development of our staff will be needlessly restricted.

¹ Alexis Wallace and Patricia Berest are faculty members at Proviso West High School who were quoted in the New York Times Magazine article, "We're all racist new"

We will all suffer because of these restrictions and limitations...."

I would like to make a second point about the article. For those without experience, that is to say firsthand experience, with our school and our staff, for those who know our school only through rumor and innuendo, the article is unrepresentative of the full reality of Proviso West. The author indicated the article is about 9,800 words which were selected from several times that number which were written for the piece. I am disappointed that those words selected did not speak to the district, faculty, and staff inservice initiatives in the area of cultural diversity in 1992 [and] 1993.... [and] to the feelings of white parents who have constructive things to say about our school....

I do have a couple of other concerns about the article. The first is any implication in the article which casts any doubt about the interest of our students in learning or the intellectual capacity of our students in general, or African American students in particular. That is the kind of garbage . . . to which people of color are subjected on a very regular basis. And whether it is Jencks or Coleman or Murray and Hernstein or other pseudo-scientific approaches, they do great harm. . . .

The second is an implication that one might draw from the article, perhaps from the title, when a large concentration of white residents leave a community, that somehow, the community is thereby flawed. I do not see our parents as flawed. I clearly do not see our students as flawed. . . .

Racism, bias, and discrimination are not produced at the school. They are functions operating in our society at large that can be played out in any private or public enterprise or institution, including private or public schools. Race-based bias and troublesome behavior that springs from it is a tragic historic part of American life. There is permanence to it, as Derrick Bell so eloquently and effectively stated in his 1992 book, Faces at the Bottom of the Well.

Given that racism and bias exist in our society whether it is housing, access to capital, or employment, and given that all of us are ... influenced by it to a greater or lesser extent everyday, the challenge for our schools is to determine a means ... to rise above that which is occurring in the greater society and to create, if necessary, an island of excellence where all individuals can feel comfortable, respected, and flourish personally and intellectually. That is the challenge for our schools, for the Proviso Township high schools, and it is certainly one we are serious about.

That is not to say that denial should occur. That is not to say that our students or any affiliated with us should be insulated from the reality of racism or racial bias.

Students of color—and we can also say women—need to understand that racism exists. They will continue to confront it and they need to be equipped to handle it... We should not put blinders on and present everything as okay. All students and adults, regardless of race need to understand that denial of opportunity to any diminishes us all....

A second point here is that we have worked very hard to diversify our faculty and staff. This past year, one-third of the certified staff hired for our district are people of color. . . . It is important for us to continue ongoing efforts in terms of textbooks and materials. All of our students need to see themselves effectively and properly depicted. . . .

In summary we are not at all in a position where we can dismiss the information in the New York Times Magazine article. At the same time I think it is important for us through this kind of a forum . . . to enhance the picture [of us] so people can understand more fully what we are about in the Proviso West Township High School.²

Leo Banks, chair of the board of education:

We have over 10 towns feeding into the two high schools and 6 or 7 feeder districts. We have a diverse [population] within Proviso Township high schools and to that end I do not think that we are any less or any

² Transcript of testimony before the Illinois Advisory Committee to the U.S. Commission on Civil Rights, community forum, Nov. 30 and Dec. 1, 1994, Hillside, IL., pp. 274–85 (hereafter referred to as Transcript).

more subject to any race views or tensions than any other community in the country

I think we have a unique position being west of Chicago and we have inherited migration from the city. And to that end we are doing our best to work with the diverse communities, communities that feed into our high schools.

Cur highest priority is the students, to make sure the students feel comfortable, are supported, and get the best possible education that we can deliver. Second to that is making sure that our staff is supported...

Right now about 35 percent of our students...go on to college and they are doing well. I am a personal example of that. I have two daughters who have graduated from this high school and they are doing well. They have had no problems at the larger or smaller universities or colleges.

Our high schools traditionally have been rated very high within the country and within Illinois. In recent years we seemed to have tapered off and to have not pushed forward as we would have hoped, but we know that and we think we are right at the measure. We recognize that improvements are always needed and we are always looking for ways to get and define those improvements.

C Rebecca Montova, principal

Proviso West is a suburban high school with a student enrollment of 2,398 as of October 1, 1994. The demographic breakdown is 56 /1 percent African American, or 1,360 students; 23.64 percent Hispanic, or 567 students; 16.35 percent white, or 392 students; 3.29 percent Asians, or 79 students. As our demographics indicate, we are a culturally rich high school. Our diversity is not only in our heritage and background, but includes economic status and academic skills.

A few years ago the school could not be described as culturally rich. However, it could be described as uniformly white . Margaret Meade once wrote, "The pattern of the future is change." All we have to do is look around us to see the truth in that statement.

TABLE 15 Faculty and staff Participants

Patrick Ambrose	Social Studies
Daniel Batka	Physical Education
Patricia Berent	English
Dale Crawford	Asst Superintendent
Douglas Deuchler	Library
Rose Hampton	Guidance
Darrell Howard	English
Connie Jensen	Guidance
Mona Johnson	School Social Worker
Beverty Malone	Guidance
Carol Radkiewicz	English
Robert Rosignolo	Business Education
Kathy Schlatter	Guidance
Gail Suffredin	Special Education
Debra Thomas	Special Education
Linda Thompson	Paraprofessional
Alexis Palm-Wallace	English

The schools cannot exist without the support of the public, and they are not going to support something which they perceive as unproductive. . . . I am proud of our students and of our staff and we will continue to make every attempt to live up to our motto at Proviso West High School, "Nothing But the Best."

Faculty and Staff

Patrick Ambrose, 2 years at Proviso West:

I have not seen any obvious, blatant prejudice [or] discrimination based upon any ethnic, religious background by faculty and, really, by the students either.

I read the Times article and I thought that (the author) took the racial part of Proviso, meaning the diversity, and he smeared that. We were under the impression in the assembly he was supposed to write a piece about Proviso West as a typical high school. I wondered if he [Bissinger] was even here. . . . He focused in on one small piece and stretched the truth, it seems to me, to sell papers. He didn't say anything about any other things; he just concentrated on that issue. And he

³ Find., pp 450-53

⁴ Ind. pp R-A

might think this is a big issue, but I have never seen it as a big issue around here.⁵

Daniel Batka, 6 years at Proviso West:

As far as my reaction to the [Times] article, I wondered if the person that was researching and writing the article actually was at our high school truthfully and honestly because I don't believe those things have occurred here. I was very upset to read what he had written. It was our impression that he was doing another piece besides what he actually came up with....

I am also the multicultural club sponsor here in which we try to get all races together, not only to celebrate our individual racial background, but also to come together and celebrate our oneness.... I think one of the things that a lot of the students have brought up is they do not quite understand the different cultures. They don't quite understand where the African American population comes from, and [where] the European American population comes from, and why some of the Hispanic population are a little bit more quiet and have a tendency to blend in . . . whereas some of the African American [and] European American population will take a stand.⁶

Patricia Berent, 25 years at Proviso West:

I am one of the people who was discussed in the article and I would like to verify that there were misquotations in the article. There were misperceptions in the article, and the article—as some of the students said—was very disappointing in that it only looked for something media worthy.

I am one of the people on whom there is a section on writing. My students were in the computer room and Mr. Bissinger would come around and take things directly off the screen of the computer. They were freshmen, basic students, and some of my very best students in the sense that they try the hardest.

I was quoted in the [Times] article as saying these students will never get to college or some such thing. Well I know that is not true. I know some of my students who start out in freshman basic—the lowest academic class that we have—have moved on to college. They moved up through the different classes, got the help that they needed, and improved. That is how this school is run.

Dale Crawford, 28 years at Proviso West:

The reader of the May New York Times piece and this morning's Chicago Tribune piece, unfamiliar with the day to day activity at Proviso West, may draw conclusions about Proviso West which I think are very inaccurate. Conclusions drawn from very broad generalizations, from quotations for which a specific context is not established, and from clear factual inaccuracies.

If racism is defined as a belief that race is a primary determinant of human traits and capabilities, then I am here to tell you that that belief is not rampant at Proviso West High School. . . Proviso's faculty and staff are committed to the belief that all students, regardless of their race, their ethnicity, or their economic condition, can learn all that we or any other educational organization can teach. . . .

If, however, people who use the term racism ... mean in their use of the term that problems between and among individuals and/or groups occur occasionally or frequently as a result of not understanding the values that motivate individuals or groups from different backgrounds, and not knowing the appropriate strategies to employ when confronted with that lack of understanding, then I am here to tell you that at Proviso Township High School such misunderstandings have occurred. ... I do not offer that statement as an excuse or as a rationalization, but as an unfortunate human reality; a reality that this school district has been attempting to address for 28 years at least to which I can speak.

⁵ Ibid., pp. 136-37.

⁶ Ibid., pp. 136-37 and 140-41.

⁷ Ibid., pp. 108-09. "Basic" refers to Chapter 1 educational programs for the educationally disadvantaged. Students in a basic program read 2 or more years below grade level.

⁸ Crawford is referring to an article in the Chicago Tribune, "Proviso West Focus of U.S. Racism Probe," Nov. 30, 1994, regarding the Advisory Committee's meeting.

If my memory is accurate, it was in the fall of 1970, that I sat in this same little theater at a general faculty meeting of the Proviso West staff and listened to a social studies teacher present a new curriculum for a course that had been written the previous summer entitled, "Afro American History." At that time there were no African American students at Proviso West High School. There were few African American teachers.9

Douglas Deuchler, 10 years at Proviso West:

I have seen an incredible number of people who have valiantly and very positively worked for positive—dealing with a lot of the issues that we have been discussing. There has always been a minority of people who just do not feel ownership of our students.

We have talked about cultural diversity. Some people do not see it that way. They see it as a sinking ship as each year we have got an increasing number of African American students. They perceived that as the end of Proviso, and I actually stopped going to one of the lounges because the gentleman in there kept referring to the urban barbarians and some of the animal behavior they would discuss was very offensive to me.

Initially, 10 years ago when our black enrollment was small... there were people who would admit to being totally color blind and the old thing about "I don't see [color]".... Rather than admitting that there might be problems here, they would say there is no diversity....

The [African American] boys are often perceived as a problem to be dealt with. Nobody cherishes them, nobody. A minority of people deal with them in a positive way at times. 10

Rose Hampton, 12 years at Proviso West:

When [Bissinger] was here he asked me a question about racism and it was off the record and this is what I told him. I said, to be honest, I don't feel racism, but I feel people ignorant to the fact of multiculturalism.

When we were going through the inservice [on multiculturalism] and the school tried to get teachers to go, people stonewalled. They did not want to hear it.... When we hold ourselves off we do these kids an injustice.... We do need somebody to help us get this stuff together here.

The teachers here are good teachers. I have seen kids come here and do not know anything and leave here with a 25 on the ACT. So there are good teachers here, but they do not and they will not learn unless they ask somebody for help and I am here to say we need help and I am going to be honest about it.

I was raised in an all-black neighborhood, all-black schools. I never lived around white or Caucasian people or Asian people. I don't know anything about them. But when I came here, I had an open mind. I used to walk through these halls and speak to people and I still do today. And they come to the mailboxes and they won't even open their mouths. And you're telling me there's not a problem. I want to know why.¹¹

Darrell Howard, 1 year at Proviso West:

I was a student here from 1985 to 1989.... From what I have seen in my time here as both a student and a staff member, the staff here is not quite representative of the student population. There are a handful of black teachers here. I attribute that in large part to the lack of black education majors at colleges and universities.

I know Proviso is active in trying to recruit minority teachers. I'm a product of their recruitment.... There are other things that Proviso could do as well, things such as education for the teachers. They are doing a lot of that [but it is] more on the teacher whether they elect to take courses. With the staff being as young as it is right now, I believe that it will change. 12

Connie Jensen, 9 years at Proviso West:

When I applied here I was fully aware that this was a culturally diverse school. I don't think I would want to work in a district high school that was not diverse. It is

⁹ Transcript, pp. 147–49.

¹⁰ Ibid., pp. 365-36.

¹¹ Ibid., pp. 125-26.

¹² Ibid., pp. 106-07.

exciting to work with all kinds of different people and I think, in general, the kids and the staff get along quite well. 13

Mona Johnson, 16 years at Proviso West:

I have been here 16 years, so I have seen no diversity, as well as diversity.... When I first arrived here, as a minority or African American, it was very difficult to be in a setting where it was predominantly white.... I guess what really disturbed me was that if you would go to the mailbox and would speak to someons, they would not part their lips....

I have had an experience where one white student came to me very upset because whites were becoming the minority. It was very difficult for him to be in an environment where [he] was walking through the halls and [felt like he] had no ownership. And I had an opportunity to speak to an African American who expressed the concern [of] being ignored in the classroom. What he said in the classroom was not considered as assertiveness or debatable, but militant.... And I have had Hispanics saying that the [school] really didn't even acknowledge their culture....

I have seen situations where our African American [students] have had to be arrested here in our building with handcuffs and taken out. Now, maybe that was procedure, but for an African American that was a form of slavery, which they in turn felt that was a bias.... So many times, we, as staff members, enforce policy and discipline [and] do not see what it is doing to the culture.... We lose that. That is why a lot of minorities feel they have been wronged in some ways. 14

Beverly Malone, 8 years at Proviso West:

Proviso is a very supportive setting to me. I've been very happy here.... People are helpful to me here and as a black female I don't feel that anyone has ever been unfair to me. If I have had problems, I feel that there has always been someone here that I can go to to get support and to get my needs met....

What I see here is basically what I see in society. There are people that do not get along here. But that is based on the fact that maybe that is a personality type situation and not necessarily just because it is a racial situation.... But I think everybody feels as though they can express their opinion here and when they do, I think that that information that they give is taken and is done constructively.....

Sometimes I feel that maybe if we had more money that would help us to be able to do other things if we had more time. It is not so much the racial component; sometimes we just don't have enough time to do a lot of things we would like to do....

I think we spend more time [here] trying to ameliorate all of the negative publicity that we get, which we get a lot of. People want to hear more of the bad things than they want to hear the good things.¹⁵

Carol Radkiewicz, 30 years at Proviso West:

The school has changed drastically in the [last] 30 years, but as far as I am concerned, the school has changed only for the better. . . I was absolutely appalled, humiliated, hurt, and very angered when I read the [New York Times Magazine] article. I think it is slanted, what we call yellow journalism, very sensationalized and very untrue. It not only hurt me, it hurt my other teachers. It has hurt my community and it hurt my students. The article is all slanted.

I spoke with [Mr. Bissinger] for about 8 hours of time. What I said the gentleman did not agree with, so it does not appear anywhere in the article. He was very clever. He looked at me and said, "Well, what do you think about the blacks, you know, do you really think that they are less academic than the whites?" I said, "They are not any less in anything." He said, "Well don't you think they really are not as able?" [I replied that] often they are more capable. A person's viewpoint that did not agree with what he was looking for at our school never appeared in the article.

I relooked at my school. Is my school dirty? Are there dirty handkerchiefs hanging out there? Are there steel

¹³ Ibid., p. 338.

¹⁴ Ibid, pp. 359-60, 362-64.

¹⁵ Ibid., pp. 338-39, 340, and 345.

doors, etc.? [The article] is totally incorrect. Our school has adjusted well to the change that has occurred. In my English classes I expect three times as much from the students now, as from 30 years ago. I am very pleased to say that there is nothing but positive changes, and this article has been absolutely untrue. 16

Robert Rosignolo, 23 years at Proviso West:

I started at Proviso West in 1972. The school was all white [then]. I enjoyed my experience then and I am still here and I still enjoy my experience. I deal with people as individuals and I try to do the best I can and that is the way I approach life in dealing with anyone....

The problem is the perception in the community, and what the perception is that is the reality. And we have tried to convince people if you feel that there are certain things wrong with Proviso West, come and see for yourself and see what it is like at Proviso West and make your own judgments. But don't listen to what other people say. That has hurt us because the reality is we are a functioning school which is trying to do the best job we can, and there are a lot of good things going on here. But there is a lot of reaction in the community that those things are not happening and that has hurt us.¹⁷

Kathy Schlatter, 22 years at Proviso West:

I was not quoted in the [Times] article because I avoided Mr. Bissinger totally. I must say once I read the article, I was not sorry.

I truly enjoy [Proviso]. I enjoy all of my students, the English department I was a part of [and] now the counseling department. I feel that there are so many people here working very, very hard for all of our students and there are some [students] who are taking tremendous advantage of that and there are those who are not....

Those people who choose to send their children elsewhere, if they think that they are getting a better education in the district, are absolutely incorrect, in my opinion. There is no question in my mind that a student here can get the best education possible, certainly districtwide. Why would they go somewhere else? Is there better discipline there? I am sorry, I do not really think that we have let go of discipline here. 18

Gail Suffredin, 9 years at Proviso West:

I can honestly say that I feel that the students are being served well. There are always going to be, in any given group of people, people who are tolerant, people who are accepting, people who are understanding. And there are always going to be a few people that are not as tolerant, accepting, and understanding. . . . The programs here are excellent. . . . Unfortunately there are [some] students in our school that are not as motivated [but] I think the teachers as a whole do their best to motivate the students. 19

Debra Thomas, 4 years at Proviso West:

Not only do I teach here, but I live in the district also. As a taxpayer and a teacher I have some real concerns that many of our teachers have lost sight of the reason of going to school to become a teacher. I feel many of them fail to motivate many of our kids and then they get labeled special ed or I end up having them in my classroom, because I am in the special education department....

I agree . . . that we need to try and recruit more minority teachers, as African Americans and Hispanics being the makeup of our population. However, I don't feel that only African American teachers can teach African American children.²⁰

Linda Thompson, 4 years at Proviso West:

When I began working [in this district] I was a white female in a predominantly African American school

¹⁶ Ibid., pp. 97-98.

¹⁷ Ibid., pp. 380, 382-83.

¹⁸ Ibid., pp. 109-11, 122-23.

¹⁹ Ibid., pp. 348 and 353.

²⁰ Ibid., pp. 370-72.

and it was a new experience for me. It turned out to be probably one of the best experiences of my life.... I was able to get rid of a lot of perceptions that I thought I did not have. I did not think I had these ideas and then I found out that, really, I did, and as I got to know the students and I got to know the parents, it was a good experience.... What I have seen here is that racism is a perception. We often think that we are not prejudiced, but all of us are to some degree....

The majority of our paraprofessional staff is African American at this time. We do have white paraprofessionals. Our secretarial staff is predominantly white female. That is changing somewhat, we do have Hispanic and African American secretaries now.²¹

Alexis Palm-Wallace, 15 years at Proviso West:

I would honestly say that 95 percent of the young people [at Proviso] have learned to peacefully coexist, either by tolerating, accepting, or even assimilating each other interculturally and intraculturally.

I respect a great majority of the individual [teachers at Proviso West] because I feel that they truly and sincerely serve the student population in earnest. [But] I cannot blind myself to the certain individuals who are biased, self-serving, self-righteous, and ignorant. As the ethnicity of the school has changed, I have seen the proliferation of rules, rules not to evoke change in behavior or to teach appropriate behavior, but to serve the lack of desire of these certain individuals . . . to deal with students from differing backgrounds. . . . We have an excellent school with superior academics, but we are not in a perfect world. And like people are flawed, so too does this school have flaws. 22

²¹ Ibid., pp. 378 and 388.

²² Ibid., pp. 349-51.

Pert 5

Parent and Community Comments

pinions regarding the New York Times Magazine article and the educational opportunities at Proviso West were obtained from parents and individuals in the community. Fourteen parents spoke at the meeting, eight were invited by the Advisory Committee; their participation was solicited because of their active role in the school. One parent followed her testime... y with a written comment, and one parent had been quoted in the Times article.

Eight individuals from the community provided comments to the Advisory Committee. Four had been invited to testify, the invitations made through the local chapter of the NAACP and another local group lobbying the school administration for more minority teachers. Two spoke at the public session; two other individuals submitted written comments, one being a letter sont to the editor of the New York Times Magazine rebutting the Provise article.

The parents generally rebutted the factual content and the tone of the New York Times Magazine article, and extolled the diversity of Proviso West and the excellent educational opportunities available for their children. Community opinion was mixed, and included other concerns such as allegations of unfair school discipline and discriminatory employment practices. Comments from every parent and individual in the community who spoke or wrote to the Advisory Committee are included.

Parents

Joseph D. Beilner*:

I am a parent and I am also a vice president of the Proviso West booster club. I have a daughter that graduated from [this] school 2 years ago and is a sophomore

Parents Joseph D. Beilner Gene Belmonte Janice Bennett¹ Terry Blaine Cynthie Breunlin² J.B. Carr Michael Luck Mary Murphy McArthur Robinson Andree Routen Richard Ryan Renate Schulz

James E. Smith, Jr.

Lucy Smith

Community
Susan Gill
Lester Grant
Latressa Hodges-Lumpkin
Garland Robeson
Linda Barney-St. Martin²
Water Sally, III²
John Thompson
Valerie Voss

at the University of Illinois in Champaign. I have also two daughters here, a senior and a sophomore.

The gentleman that wrote the article spent roughly 2w hours at my home and spoke to all three of my daughters . . . my wife and myself. [We] talked 15 percent of the time about concerns [at Proviso West] that are concerns in every major high school in the United States, [and we] talked 86 percent of the time about good programs that happen at [Proviso West]. . . .

. . . I talked to this gentlemen, and I am almost like Bobby Knight, the backetball coach at the University of

Quoted in the New York Times Magazine article.

² Submitted written comments.

Indiana, [in that] I hate newspapers, I hate the media. Looking at the [Times] article [angered] me.... I don't understand where he came from.

The newspapers, the journalists, and the media just take things out of context. [This guy] set some of the kids up in this school. "May I quote you on that?" [he asked], and the second part of the sentence, that's what he doted on. The kids learned from that last year. It is unfortunate that some people around our town do not wake up to the value of education that they can receive here at this school.²

Gene Belmonte:

My oldest son was a student from 1984 to 1989 at Proviso West. My younger son is currently a student now. I am a former school board member of school district 92½ in Westchester, a feeder district to this school district. I am a PTO member there. Uniquely enough I am also a former newspaper owner and newspaper editor of the Westchester News....

Magazine article—as many other people's reactions to it—is almost humorous, but it is actually more tragic than it is humorous. . . It is very difficult to fight a national powerhouse and a member of the media, a super media as the New York Times, especially when there [are] sensational aspects. . . .

... The article was inflammatory, and the reactions to it unfortunately fan the flames on both ends of the spectrum, the racism as it is. And that racism does exist, both in the black and white community.

The fact that you are investigating racism at Proviso West is incredibly ironic. The reason that my sons go to school here is because it is a multicultural institution. This is one of the few places in the western suburbs and in the Chicago suburban area in which you can experience a multicultural experience while going to high school. The majority of schools are tremendously black or tremendously white. . . .

...The fact that [Proviso West] has to defend itself and go through this is ironic. It is not void of the problems that exist in all facets of American education, but it is a place that works very hard at being good and fair to students and educating them. . . . Problems do exist [but] I don't think it exists here to any degree that merits this kind of treatment for them. . . . 3

Janice Bennett*:

I have not experienced any problems at this school. My daughter did not attend this school the first year; she went to a Lutheran high school because I listened to what other people had to say instead of coming up here myself to see what the school was like. My daughter has been here since she was a sophomore and she has grown tremendously at this school....

... [The New York Times Magazine article] was not an accurate reflection of what happens [here]. In fact I called [Bissinger] after the fact because I was very upset. We were so excited. We thought this is going to be great for the school. When I picked up the newspaper and I started reading it, I was hurt. And I was very angry... at [Bissinger].

He came into my home and misrepresented himself. He sat at my dining room table, ate my cake, drank my coffee, and he put a bunch of lies in the newspaper. . . . He takes one thing and puts it in the paper to make it misconstrue everything you say. So as far as what he said, his credibility was lost and the paper was not accurate. 4

Terry Blaine*:

I am a resident of Westchester. I currently have a senior daughter here. My oldest daughter graduated in 1990 from here. I graduated in 1961 from here. So I have been in Westchester for a while. I have been an officer in the PTO for several years. I was in the original group that formed the PTO. I was their first president....

² Transcript of the Illinois Advisory Committee to the U.S. Commission on Civil Rights, community forum, Nov. 30 and Dec. 1, 1994, Hillside, IL, pp. 160-61 (hereafter referred to as Transcript).

³ Ibid., pp. 218-21.

⁴ Ibid., pp. 174 and 176.

... [My daughters] have never come home and complained of any racial problems. So I won't go so far as to say that are none here because I am sure there are. There's some where I work. It's everywhere, but it does not seem to affect students and the education that they are getting [at Proviso West]....

....[An] example [of media misreporting] would be the news last night when they covered this Committee being here. The closing remarks of the reporter was that there was, I believe, 5 students from Proviso West and 16 from Proviso East that were expelled because they were found with guns and harassing teachers, physically abusing teachers. What [the reporter] did not tell the people and the general public was that the student did not get in the school with guns. But now the perception is everybody in this school is carrying a gun. That is the type of press that really hurts. They don't tell quite the whole story. And I think those are the things that really get us in a bind.⁵

Cynthia Breunlin:

We too have been disturbed by the powerful effects of the New York Times article. The glare and the spotlight on one suburban school made it impossible to pass the referendum, difficult for teachers to hold on to their enthusiasm, and confusing for young people.

Raysm is prejudice with power to harm. It means to me that only those with power should be held accountable. Random slights and insults by those with prejudice can be corrected if racism is addressed. Unfortunately, the wider community also bears responsibility for the racism directed towards Proviso West. The myths and misperceptions swirl around the young people.

West Cook leaders visited Proviso West Oct. 18, 1994. We were pleased to tour the building and visit class-rooms. I wrote a great essay to the local paper to encourage support for the district. . . We offer encouragement and support for district 209. We appreciate the efforts of the board and Dr. Eversley. Diversity is our future and we need to address the challenge of diversity, including negative attitudes and

institutional practices. Only by working together can we provide models for young people in education.⁶

J.B. Carr*:

I personally have not had any bad experiences [at Proviso West] and my child has not related any [bad] experience that she has had as far as race is concerned. I am active here at the achool . . . so I am around the kids a lot. I work with the concession stands when the kids are loose and out in the open and [away] from parent [control]. I have not seen any racial confrontations at all. I have heard that there are gangs . . . in the school, but my child has not expressed any situations with them. So as far as I am concerned on a personal basis, most races . . . get along well in and out of the classroom. . . .

... When I heard about the [New York Times Magazine article], I asked my daughter her opinion since she is up here and she has contact with most of the teachers. She did not agree with how the article came out. So based on that ... it was another media newsletter, and I teach my daughter to only believe half of what she reads if she is not there and cannot verify it. . . .

expressed a concern of a racial nature. . . . You do have parents come and express that they feel that a decision was made or based on the race . . . I feel from a personal experience that they are dealt with fairly, because if they have any question, the assistant principal who is normally in charge of [discipiine] situations, . . . gives an answer, some type of response. Now whether they are happy with that response [is another matter], because I have a child here who has been expelled for a situation. And I did not feel that it was handled properly, not because of race, but just because of the rules in general.

Michael Lusk*:

I am from the community of Westchester. I have been in the community for 15 years. I am currently a school board member for Westchester district 921/2. I have one

⁵ Ibid., pp. 409, 410, and 416-17.

⁶ Letter to the Illinois Advisory Committee to the U.S. Commission on Civil Rights, Dec. 20, 1994, Midwestern Regional Office, USCCR, files.

⁷ Transcript, pp. 173-74, 175, 179, and 181-82.

child in school [here] as a freshman, and I had one child graduate last year, who is now attending the University of Iowa. . . .

... I spent an hour and a half with [Bissinger] on the phone on a Sunday night and he did not mention my name at all [in the article]. Every time I talked to him and answered questions for him I was very positive about Proviso West. But in looking back at the article and having heard comments from other people, I could definitely see some type of a pattern. Every time I said something positive, he would bring up a negative and he would pounce on that. When I would move on, he would go back to that and say. "But isn't there something else you know?" "Why are they doing this?" In reviewing, I can definitely see he was not looking for the positives, he was looking for sensationalism. And that is what he got. . . .

... I am very pro Proviso West. [But] I would love to see some different faces at some of the PTO functions instead of seeing the same faces for the last 5 years. Parental involvement at this school is very lacking. We keep pounding on people to participate. We have a PTO meeting four times a year and we only get 40 or 50 people. Sometimes we get 70 [and] I almost have a heart failure. But that is the type of attitude that comes from this type of socioeconomic reality in this school [and] sometimes it is hard. . . .

participate in the PTO. I have been here 5 years. And I walk through the halls, and it amazes me the teachers that I haven't seen, and they have been teachers here for 20 years. Which goes to show that the teachers here need a kick too.⁸

Mary Murphy:

I have six children; three have graduated from Proviso [West], two are now attending, and I have one in grade school.... I asked each of my children before I came here today, "Did you ever have racial problems? Are you having any now?" They have not.

My son was in Mr. Patterson's class last year. He said what was in the newspapers [New York Times Magazine article] was not what he saw in that class-

... None of us are perfect, but I believe that here at Proviso we are trying. We need more parents involved in cooperation. . . . Students come in this door and teachers are expected to work miracles. It is a cooperative effort between parents, teachers, students, and the community.

I love Proviso West. My daughter turned down a scholarship to go to IHM [Immaculate Heart of Mary] to come to Proviso. I have a son who is a National Merit and Presidential Scholar winner. He is graduating from Illinois State University this year with a degree in chemistry. All of my children have gotten a wonderful education here. I don't have complaints about the school. Yes, I believe it can be better, and I want to roll up my sleeves and help.⁹

McArthur Robinson*:

I have two children here, one that graduated from here, and I am involved with a couple of activities here. . . . We have gotten a lot of negative publicity from the things that were stated in the [New York Times Magazine] newspaper article and on the whole I think this is a good school. I think kids get a good education over here and the ones that really want it, they get it because it is here. . . .

Anywhere you go there can be a racial problem, but I don't think it is what it is made out to be here. I've made a good rapport here. I've come to parent-teacher meetings with my kids to get their grades and everybody that I have met so far I have had good rapport with....

... I've [heard people] speaking about the noise level here at this school. And I've been here during when kids are moving from class to class and the noise level is above normal. But the kids we're raising today are much different than 20 or 30 years ago, for whatever

room. They love Mr. Patterson; he treated them well. They got along with the black students. He said there were times when they kind of paired off because they had more in common with each other, just didn't know each other that well. But we have not had a problem, and they all felt very badly about the article when it came out. . . .

⁸ Ibid., 432-33, 437, and 438.

⁹ Ibid., pp. 128-29

reason. We can cite a lot of reasons. It's kind of hard to checipline kids at home now.... You can't even, if you see a kid that's doing something wrong now, you can't even say, "Son, I don't think you should do that." I've said that to a couple of children that I've seen doing some wrong things and I wouldn't care to say the language what they told me. This is the way it is now.... 10

Andrea Routen*:

Proviso [West] has gotten a bad rap with a lot of negative publicity and it is hurting the community. It is hurting the quality of teachers we have. And it is definitely hurting the students. There may be a lot of students that could or could not decide to come to Proviso West in the future. . . .

.... We have a lot of quality students currently living in the area and if those students and their parents decide to uproot and move in different areas, the reflection of some of the incoming students may or may not have the same potential as those that are currently leaving.... So what we have is an unequal balance in the educational system; it is a little lopsided. The quality of the education in some areas surely differs from other areas. There is no doubt... what is coming [into this district] sometimes creates a negative for the people that are deciding to stay....

... The students that we have in our school right now, if they see that this is the primary issue, then what do they have to look forward to. We have the opinion of one person creating total uproar in our community and it hurts, it really does. I have not seen any direct racial problems. I am not saying that they don't exist, because they probably do, but I have not been directly involved in any racial situations. 11

Richard Ryan*:

I think if you start with the premise that racism, however you may want to define it . . . , exists generally in society, then it would be naive of us to assume in a multicultural school such as Proviso West that it does not also exist. We have to accept that fact. Do I personally think it is a dominant theme around here? No, I do not. Certainly there is going to be incidents among the students, between the faculty, and between the faculty and the students to one degree or another.... Anytime you get elements between people who do not know each other, there is going to be misunderstanding.... That is typical of what may happen around here.

I have a daughter here who is a senior. I have a daughter who graduated from this school, both of whom are very favorable about their education here. Have there been incidents that they have reported to me? Certainly there have been, but no different than what you would find at any other school, no matter what their racial maksup. . . .

... I think the school gets a bad rap from the media, both locally and nationally, because of the ignorance of the facts. They never have checked us out around here. We talk among ourselves... that most of the people in the area that rap Proviso West have never set foot in the school. They don't know what this school is like. They have no basis to make a criticism when they have no idea what happens.

I think the media in this area is extremely biased against Proviso West, not withstanding any racial issues or prejudicial issues. If you look at the local newspapers, whether they be the Suburban Life, the Westchester News, or whatever it happens to be, all the articles you see in there are about other areas, other schools. And particularly the Suburban Life is extremely prejudiced in favor of Lyons Township. That is all they write about. . . You never see anything about Proviso West. 12

Renate Schulz:

My daughter was grabbed by a boy. He happened to be a black boy. But she is muscular and she pushed him against the wall. And he said, "Oh, I didn't know she was that strong." So nothing bad came out of it; my daughter wasn't expelled and he wasn't expelled.

¹⁰ Ibid., pp. 419-20 and 421-22.

¹¹ Ibid., pp. 410-11 and 412-13.

¹² Ihd., pp. 160-61 and 167-68.

Maybe students don't know they are not supposed to touch each other. They are not supposed to. My other daughter, she is out of school now, at that time her hair was still quite blond and long. And some students, black students, tried to touch it. Now I don't know what is so special about it, but white students never touched her, or kidded her, or anything. Maybe the kids just should be told not to touch somebody because that is not proper. 12

James E. Smith, Jr. *:

I am a parent of a senior here. I am president of the PTO. I am involved with the booster club, and I am a graduate of Proviso West. ... Regarding the [New York Times Magazine] article, I was not happy with it. I didn't think it portrayed Proviso as it is. It focused only on black and white, which I think ignores the diversity that is here at the school.

I think you have seen statistics on our student population which is probably 57 percent African American, [around] 25 percent Hispanic. 18 percent white, and 3 percent other. So "other" is a mixture of Asian, Indian, Filipino, and many others. There is a program for students who have a first language other than English. At last report, they had—I think—16 languages being spoken at the Proviso Township district other than English as a first language. Those students come here to Proviso West. . . . The article did not address any of that. . . . I don't know a school that has this kind of diversity that is not a unit school district. Proviso West has, minimally, six feeder districts. And to assimilate these people and to provide them with the education they do, I think is quice excellent. . . .

... The irritation that we have within the parentteacher's organization is that we seem to always be defending the school from things that are inaccurate... As parents we are more concerned than the general population as to the safety of the students; whether there's gangs, whether there's drugs... My wife and I have been here during the school day. It is a safe place, it is a good place.... ... We were happy when my son chose to come here. He chose to come here. We gave him the opportunity to look at the schools in the area and he chose to come here and that is one of the reasons we got involved. We heard the negative stories, [but] we're here. 14

Lucy Smith:

I have a son who is a senior here. I've been very involved in the PTO organization and I've been very happy with the education my son has received. He particularly chose to come to this school. We gave him an opportunity to go wherever he wanted, and he chose to come here. We looked at private schools and other public schools, but he wanted to come to Proviso and he wanted to be in a community that was multicultural and learn about different kinds of people....

... The PTO organization is a black and white group. We get along very well and we are looking to improve the school as best we can. It seems like the past years I have been involved, half the time we spend trying to figure out how we .an convince people that all the things that they say in the paper are not true(it is not) happening here.

The basic attitude of the PTO about all the bed publicity is to say that we don't feel a lot of it deserves to be. You give credence to something if you have to go and defend it and say that it is not the truth. So a lot of times we don't want to say things and defend every time somebody says something outrageous. 15

Community

Susan Gill 16:

My children are mixed. My husband is black and I am white.... My children, in their unique type of situation, would experience negative things and positive things from both black people and white people.... I never felt that the way a teacher would react to my children had to determine how they did in that class. They had to determine how they did in that class....

¹³ Ibid., p 236.

¹⁴ Ibed., pp. 419, 422, 424, and 426-27.

¹⁵ Ibid., pp. 212-13 and 217.

¹⁶ Susan Gill is the mother of biracial children who graduated from Proviso West in 1992.

... My biggest problem at Proviso West during the time my children were here—and you have to understand that it was a different school board and a different administration and even different guidance counselors—was with the guidance department. Black children were basically being told—not now—at that time, not to pursue higher education. Even my own children who happen to be straight A students were told not to bother going any further than Triton College. ... What I did about that was I went to ... the head of the guidance department ... and I told them that the guidance counselors would not counsel my children, do not allow them to do that, ... and that was taken care of. ...

... During the time I was a parent here there was a big transition going on with the board of education. The old guard who had been in power for 30 years and were still living in the fifties basically were retiring and they were leaving and we had new people. . . . So I think there is a lot of potential in Proviso West, the Proviso West High School District, tremendous potential. 17

Lester Grant

I contend that [Proviso West] is not that bad as it stands today. I happen to serve on one of the boards of education that send children over to West and I was highly insulted when I read and heard some of the comments that were made in the news media coming from even teachers here at West. I hope the individuals at West that made those statements have found themselves another place to teach where they feel comfortable and not interrupt and hinder African American and other students. The whole achool will suffer if we have cancers like that in the school that prevent young people from getting an education.

As far as dealing with one of the communities that wants to go off on their own, I hope the citizens will

consider what they are doing before they follow those guidelines in those footsteps in people that have taken them down the wrong path. If contend that we all as citizens owe it to all of our children to see that they all get an education. If you are going to live in this world today, you are going to have to deal with diversity or you will not be able to survive and prosper...

As far as knowing of any particular incident as far as discrimination, I am not aware of any. But looking at things realistically, you know that it is there, it is just observation. I guess my experience as a black person has [made] me pretty much of an expert in institutions like that. . . What I have been hearing is that people say in the past they were able to teach because they were teaching white children and white children are being taught by a white teacher. If they could do it then, what is the problem now? 19

Latressa Hodges-Lumpkin:

I am vice president of the West Suburban Proviso Chapter of the NAACP.... Our position has been that discipline is not meted out fairly. In expulsions, there were 21 this week; 5 here at West, 16 over at East. Suspensions, be they at home or in school, are not meted out fairly. We found that black and Latino students were disciplined more harshly for similar infractions than their white counterparts....²⁰

Latino, black, Asian, and Native American teachers. ... What we find time and time again is that the minorities are hired either to teach P.E. or relegated to paraprofessionals or to janitors or custodial workers and they are not in the classrooms. ²¹ We have not looked to inflame any of the situations, but to address situations and correct them because if they are

¹⁷ Transcript, pp. 394, 395-97

¹⁸ The New York Times Magazine article identified "Citizens United for Better Education, a movement founded by several Westchester residents in 1992 to explore the possibility of 'de-annexing' their town from the Proviso high school district (Times article, p. 28)."

¹⁹ Transcript, pp 193-94, 196-97

²⁰ Suspension data for the 1993-94 academic year by race and ethnicity is in part 2 of this report.

²¹ Fifty percent of faculty and administrator hires and placements at Proviso West in the last 2 academic years have been minority. In the 1993—94 academic year, 19 new faculty were hired at Proviso West. 16 white and 3 minority. In the 1994—95 academic year, 27 new faculty were hired, 23 white and 4 minority.

not addressed, if they are not put on the table, they are not going to be corrected.

I heard about the 2-day (in-service teacher) orientation. From what I have heard from some of the faculty members and teachers who participated in this, it was treated like a joke and it was not taught properly, that it was not implemented, and teachers did not come away with anything. I keep hearing a lot of multiculturalism and those have not been the things that we are finding. It

The Advisory Committee asked Hodges-Lumpkin for specific information she or the NAACP had on racial discrimination in the district.

Yandle: You mentioned some of the students and some of the research that the NAACP has done. . . . We ald you send us some of that information?

Hodges-Lumpkin: Ms. Constance [Davis] has it. We sent them [the Midwestern Regional Office] shipments of stuff. ... ²⁴

Pugh: Can we conclude on one point? Would the information you [Hodges-Lumpkin] are talking about be made a part of the record here?

Hodges-Lumpkin. I do believe so, because we did submit it. Those things have been submitted (to the Midwestern Regional Office) prior to [the meeting].

Pugh: So it will be submitted tomorrow, [because] I have not seen it....

Hodges-Lumpkin: It has already been submitted. I am sorry, do you want extra documentation? I don't know if Peter [Minarik] and Constance [Davis] are going to share those things with you.

Pugh: I don't know

Hodges-Lumpkin: I don't know what we have to do. If I need to resubmit it, it's not a problem. I'll resubmit it. 25

Garland Robeson:

I am here on behalf of Arme Bryant, local NAACP president. [The NAACP] helps all students, be they black or white, with problems they might have concerning questions about their civil rights. ... Basically it is student-teacher interactions and the negative kinds of feelings that the black students and other students get from things like discipline and attendance kinds of situations. ...

Walter Sally III*

Mr. Bissinger was not wrong in his assessment of the overall observation that racism is "alive and well" here. I heard the majority complaint by those selected to address the Commission that Proviso was getting a bad rap. I am also aware of the panel's concern for the lack of community outpouring to substantiate the allegations of racial disparity. One reason is that many

On Aug. 25, 1994, the Proviso Township High School District had a faculty institute. Much of the institute was devoted to multicultural issues. Breakout sessions included: "Cultural diversity in the workplace," "A World of Difference," "Quality relationships," "Multicultural competence," "The New York Times Magazine: Talking out the issues." Among the presenters and moderators were the Community Relations Service, U.S. Department of Justice, and the Anti-Defamation League of B'nai B'rith. The agenda of the institute is in app. 4.

²³ Transcript, pp 225-26

²⁴ Transcript, p. 245. The Midwestern Regional Office, USCCR, received four local newspaper articles from Hixiges-Lumpkin. The articles dealt with deannexation attempts by some Westchester residents.

²⁵ Ibid., pp. 249-51. Hodges-Lumpkin did not submit to the Midwestern Regional Office, USCCR, any information other than the four newspaper articles noted in footnote 24, nor did she submit any information to the Midwestern Regional Office, USCCR, or the Committee subsequent to the meeting.

²⁶ Ibid., pp. 402-03, 405, and 407-08

were not aware of the forum until that Wednesday morning's article in the paper. The local papers only advertised the vote on the tax referendum prior to that date. Many others did not want to go through what that poor unfortunate student went through for speaking up. Please find also a copy of the 1992–1993 budget proposal. Note that there isn't any monies allocated for better race, residential, or student relations. Not a priority I guess!²⁷

Sally further asserted that the Advisory Committee is exhibiting its racism by having a meeting at Proviso West High School and about Proviso West High School.

The very fact that this [Advisory] Committee is here only when West is addressed with racism, is racism at its best. I have here a 1987 article from the Maywood Herald that racism has been alive in the Proviso Township some 27 years, but when it gets to [Proviso] West, then the Commission is convened and we want to talk about it. It has bled for 27 years, and now that it is ensoaked and entrenched, and now that West is getting some of it, now we want to talk about it.

The main body of African Americans in the Proviso Township exist at [Proviso] East, but it is only when it starts being questioned over here [at Proviso West] is when they say, "Well let's go check it out and see if it is real." 28

John Thompson

I am a former employee here for almost 4 years but recently I had to resign. From my knowledge of being in the hallways I will see students running to the door and West has a policy if you are not there, we are going to shut the door. But if you shut the door when the student gets right in front of the door when the bell rings, there is no way that student can learn. Sure

some of the students don't want to be in the class, or they cut and they go to the cafeteria anyway, but it is a large majority of them that do want to be there, but they get turned away, and I know there are teachers who can youch for that.

Approximately a year ago there was an incident that occurred in the cafeteria. Me and another paraprofes simal were there. About 20 students were searched, 4 or 5 of them were young ladies. The policy is you do not search anyone until there is a dean or someone of authority on the grounds or you have permission. There was no permission given to be police counselor that we currently have on staff. He asked me to assist him in the search. I refused. He continued on with the search. When he got to the femalos, he searched them; he searched their purses and everything. The policy is you have a female paraprofessional when you are searching females.

There was one young man, he was an African American male. He refused to be searched and he expressed that to the police officer. He [the police officer] used profamity, just cursed at the kids. The other paraprofessional that was there heard it as well. So I went back to the staff to report this. Nobody did nothing.

Valene Voss:

My daughter was a student here and I took her out of the school system last year, not because she was a troubled child, but she said she just got so tired of the challenges that she faced here at this school. One occasion she was kicked out of the classroom because she questioned the teacher. She was in an English class and the teacher told her she was using black English in writing a story. Her response was that she knows the difference between black English and standard English and this is the way the people were talking in the particular story.

The teacher told her that not only

Walter Sally III to Peter Minarik, Midwestern Regional Office, USCCR, Dec. 21, 1994, Midwestern Regional Office files. Included with the letter to the Midwestern Regional Office was: 1/a copy of a petition with 308 signatures, initiated by Sally, "to demand better recruiting and hiring of minority educators for our children in the Provise schools, specifically Provise East," and (2) Notice of Substantial Evidence, from the Department of Human Rights, State of Illinois, July 23, 1991 re: Sally vs. Provise East High School. Provise East is the other high school in Provise Township High School District 209

²⁸ Ibid, p. 209. As set out in the introduction, Provise West High School came to the attention of the U.S. Commission on Civil Rights in Washington, DC, when the article about the high school appeared in the New York Times Magazine. The meeting is a solicitation of opinion on race relations and equal education apportunity at the high school. Neither the U.S. Commission on Civil Rights nor the Illinois Advisory Committee were aware that a Provise East High 3chool existed prior to this project.

²⁹ Transcript, pp 198, 201-03

was she using black English, but she didn't know how to write. . . .

make the letter M and my daughter said, "Well, I've been writing that way ever since I learned cursive." And the teacher said, "Well you don't know how to write." And so my daughter's response to her was, "[We have] something in common because sometimes when you write something on the board I don't understand either." My daughter was asked to leave the classroom, and they wanted her to do a 20-minute detention, which she refused to do. 30

Linda Barney-St. Martin³¹:

[On] Friday, June 3, 1994, I took the opportunity to visit Proviso West. . . . The visit was revealing and stimulating, and a great deal was learned about Proviso West.

The Director of Student Affairs, Mrs. Annette Karones, gave [me] a tour and I was given the opportunity to talk to the students and the majority of the Proviso staff. I

found both groups were extremely disappointed with the article.

After my tour one student asked what I thought of Proviso West now that I had a chance to see the school? My response was that I felt that the article was very inaccurate. The school was in fact a peaceful and very pleasant environment of learning.

Please note this student was not a reporter that was on the premises for 2 months, in fact she has been a student for 3 years and felt that her school was shamelessly violated by this article. Who would know better, a person who visited the school for 2 months or someone who has been attending the school for 3 years?

The picture shown of the police vehicles in the parking lot were facing Westchester, not Bellwood, as was indicated by Mr. Bissinger. I live in Voorhees, New Jersey, near Eastern High School, and police vehicles are there every day to get the traffic in and out of the school, so perhaps Mr. Bissinger was not aware of the fact that they are to control the traffic, not the students.³²

³⁰ Ibid., pp. 391-92.

³¹ Linda Barney-St. Martin, M.D., a former Ms. Black America, visited Proviso West after reading the Times article. She wrote a letter to the editor of the New York Times in the aftermath of her visit and gave permission to the Committee to publish her letter as part of the community comment.

³² Linda Barney-St. Martin, In.D., to the editor, New York Times Magazine, June 10, 1994, Midwestern Regional Office, USCCR files.

Part 6

Conclusions and Observations

The Advisory Committee conducted a community forum at Proviso West High School in response to the Commission's concern over an article about the school in the May 29, 1994, New York Times Magazine by H.G. Bissinger, "We're All Racist Now." A community forum is an activity of a State Advisory Committee designed to elicit opinions and perspectives about civil rights matters in a local area.

It is anticipated that the value of this report will be a useful update on the attitudes and conditions at a changing high school, changes and attitudes which are far more positive than expected. To this end, the words and opinions of the students, faculty, administrators, parents, and other participants are presented without comment, and remarks from every individual who spoke or wrote to the Advisory Committee are included.

In the last 20 years Proviso West High School has undergone a transition from a virtually all-white school to a majority-black school and problems have attended the change. During its 2 days at Proviso West, the Advisory Committee found and heard about a school that is functioning in a changing environment. The operation of the building is orderly, students are generally attentive, and the faculty is by and large committed to offering a good education to a culturally, ethnically, and socioeconomically diverse student population.

The administration of the school and the high school district cooperated fully with the Advisory Committee in this endeavor. The meeting was held in the theater of Proviso West High School, and most of it was conducted during school hours.

On several occasions during the 2 days, entire classes of students came with their teachers to watch the proceedings. In addition to furnishing accommodations, facilitating student and faculty participation, and providing background and demographic information, the school administration allowed its building and classes to be open and available to Advisory Committee members to tour and visit.

In examining the situation at Proviso West High School, the Advisory Committee solicited opinion on the climate of race relations and the equality of educational opportunities at Proviso West High School. To ensure that such opinion was balanced, student leaders were selected, as well as a second group of students randomly drawn from the student population. Based on what was presented, the Advisory Committee could not conclude that Proviso West is not providing equal educational opportunity to all students. On the contrary, there was no suggestion that the educational opportunity offered to whites is any better than that offered to any other racial or ethnic group. Racial tensions do exist at the school. However, to the extent that individuals stepped forward and indicated that there are some race relations problems, both between students and between teachers and students, the statements are vague and general. No specific examples and no details of racially motivated incidents, indicative of race relations problems or minority students not getting equal education opportunity, were offered to the Advisory Committee. The Advisory Committee understands how there are underlying tensions in this racially diverse school, but it did not hear specific cases or

¹ U.S. Commission on Civil Rights, Administrative Manual, AI 5-7.05. A community forum differs from a factfinding, which is the gethering of data, documents, and testimony that enables a State Advisory Committee to reach a determination of facts.

examples of overt discrimination or overt racial antagonism.

The testimony critical of race relations at Proviso West High School broke down mostly into two categories: (a) general comments by students, parents, and others that racism exists at the school as it does in the society at large, and (b) more specific criticisms (all from nonstudents) regarding the recruitment, hiring, and promotion of minority employees, especially teachers.

The Advisory Committee heard of the efforts by school and district administrators, parents, students, and teachers to support multiculturalism and to discourage discrimination at Proviso West High School. However, the Advisory Committee feels more can be done to move along this difficult road. It does seem that Proviso West has made more of a commitment than most American institutions to multiculturalism and non-discrimination, and it is therefore appropriate for it to make a superior effort to achieve the high standards it has set for itself. One particular area is increasing student and staff sensitivity to the need to work constantly at improving relations between individuals and groups of different races.

Criticisms about not hiring or promoting enough minority faculty, especially minority male teachers, were voiced by one teacher and two individuals from the community. Although no one explicitly made the point that minority students, especially boys, benefit by knowing and encountering minority teachers, the Advisory Committee believes this to be true. The high school district superintendent also appears not to be content with the number of minority teachers and has done some recruiting of minority teachers. Still, just 7 of the 39 faculty hires in the district for the last 2 academic years have been minority

(18 percent). The Advisory Committee feels that the administration should make greater efforts to hire and retain minority teachers, especially minority male teachers.

Most students and parents who testified spoke positively about Proviso West High School and the quality of race relations in the school. Most also acknowledged, some of them responding to questions from Advisory Committee members, that they are aware of racism in the school. Most of them went on to declare that they had not witnessed or experienced racism personally, and to assert that whatever racism exists at Proviso West, it is no worse than elsewhere. No one condoned racism or felt that the existence of any racism, even if limited, is acceptable. In some respects, it seems to the Advisory Committee that many in the school community hold themselves and the school to a higher standard of nondiscriminatory conduct and thinking than does the society at large.

In conclusion, the primary consideration in the provision of any education is the students. In this respect the comments offered by the students are most illuminating and perhaps most reflective of the racial and ethnic environment at the school. Their statements and other testimony raise two specific concerns about race relations and equal education opportunity at this racially and ethnically changing school.

First, it seems probable that a residual attitude of prejudice from parents and friends resides in the feeder communities to Proviso West High School.

Second, the students at Proviso West seem to have learned to tolerate and ignore prejudicial behavior at the school and among their peers as long as it is not directed towards themselves.

Appendix t

Proviso West started as a jewel of a suburban high school. But then for 20 years the school board, faced with the needs of a growing influx of black students, fretted over maintenance and patronage hiring. Whites fled, standards fell and

HEN FOST SPEED OFF he thick spine of the ove Express vey, Preview High School suggests all the impred ents of a stable place. Its low-chang beings brick, underlying for acre after acre over the fairways and greens of the old golf course in Hillands, Il., makes a seem more like a corporate campus than a reduction bigh school. lu locacion, austisti in a comfortable perch between a commenty, a shopping mall and a move theater, gives at the feel of a sanctuary scaled off from the monocone of Chicago's waters n who, with their mades stray made and four-base highways and fast-food resuuraes. In the strong earlymorning light, when the school parkung lox is empty of the rattle of cars. and students have not yet began to congregate ownide the metal doors werring to be let us there is sen thing truly monumental about Provec West a place built in the beharskeiter ruburbus-boom-sown days of the 1950's — and built to made

lasse the school there is the feel of a piace where unpowers' meany was well speak — a had of mean-lion square feet. Adjacent to the mean complex is the high enemy of the field bouse with its fast track surfaces. There is the gym with row after row of rising blanchers, and the sperkling indoor pool, where a swimmer pulls smooth, fine strokes that harely break the water.

The grounds, at first glasce, are just as lovely: well-kept wooden benches, an outdoor stage, the Sensor Circle courtyard so falled with the chirping of birds that it seuads rigged—all the shrubbery impreciably man-

WERE ALL ST NOW NOW

BY H. G. BISSINGER

H. G. Basseger is the casher of "Friday Nagis Lagiss," about the obsession of Odessa, Tex., with its high-school football team. He is corously working on a book about the factor of the American city.

^{*} Copyright ©1994 by the New York Times Company. Reprinted by permission.

entred and the expend, he can extend of Steam Cards share a se degree model due was demand by the dam of 1944. Here the resided is a sense bands demand by the dam of 1945 with an exemption that cape, "The Fernic Balange to These Who Process for the

Person Wen and in over school Proving East, Seen what is become a Proving East, Seen what is become a Strong Seen what is of the Seen and its de dame, if the see were seen, if of them is the seen where seen and the Chings, will population require from the PLAN. Registering in the 1987 way of these betweet from the willings are Uniting buildy common, Seen, See Belleved, but there were emeng the working than the reast emeng the working than the reast emeng the well of the province of the Chinese and become where makes increase or beginn the seen and among the wey IS province of the Chinese and they were the decomposited by completing the makes are all the seen and the seen working feeling into the common of the province of the the seen at the their population to speak of in 1988. The others of had their populations

It was as this on the Province (preserved pro-VEYE-as). Were you quered both in the law 1939, with a M-7 million head from a symbol Ar the defension approximate of Mer. 16, 1938, the work of the Astronom observed "Our purpose of an exhibition in the largest manches of our famous difference of the second continues in the largest manches of both the responsibilities and the heads which comes or damp because they are Astronom on damp because they are Astronom and fore.

Law on, it the prosth-card-cards sown arranged by voluments, seven through paper over to altain the place there are defined buth. And what a place it was, with thereto eating place and arranging seven as any speed, with meeting the the through the and the cards to are did not arranged through the area of the speeds seven of Saus, before these through the arranged through the seven through the character of the arranged through the seven through the s

Thirty-on years here, these would be home above invisible. The word is self there, but the blac of inthe infahrd, denser become been. Sendered from many day Sendere Could many company of the Sendered Sen

Over the past density. Person West has expensive of consequence of the makes past 1990 of the makes of the makes were from law-amount from law

femilie. This year, the figure has almost deathful, to 39 persons. In 1973-74, the school, with roughly 4,550 readoms, we have then 1 persons for the season of 40 persons which. Here, 20 years here, there are only 2,350 makes, of when roughly 50 persons to black, 22 persons Leries, 10 persons Leries, 20 persons to black, 22 persons for history and the balance Aries.

Berran 190 and 190, the paper hairs of the sevening cheest seath 12 persons, so 152,000. Persons or authorises we find from the seath of the paper of the table of table

Same whites who want to Provice West in the 1999's would notes, over used their shallows when. They say the best shang to do would be so obsee in said of a cell to a hig experience. They describe in as a "jal" and complete short how the phase is hilling their property values. When eached to shield of ways to When eached to shield of ways to of the school, they came up with idea the morel discesses and hall proceps with E-P paths weating down.

"I would say with soul denous, I would had a light hit hasse," says one power, Assemb Leiblik. "For sowing the work is a set and po your with an AL-O." (Subset and power in the lines only one denoting incident on Province West proposely in the history of the subset.) As for the E-O days, Leiblik our the idea on plan for exceptable "Do bring the days in. If nothing's beginning the worst the ways."

Labelo be desired the lar so children, both of when on our one to you, will one go so Provine West then the should be the whole into all parties of any most on her saint fits worsel as the conditions of Chinese United for Breez Edon time, a movement founded by some place the paradilary of "Variances and the paradilary desire was founded by the paradilary of "Variances and the paradilary desire the hardy-based desire. The idea was a specially beauty American Calley Paradilary (ACCE) seems, the see paradilary the paradilary of the said the

Southwen Airlines, is now weeking with the group on months plan of which Wandpury and packaps and or two other accessing world separate from Provine West and oron their own school dispries.

Thems Walder, a larger from Worshour who is among these studying the pare dentice idea, or phins the impress behind in the good of medium from Permissers are simply so larger encapsable was the made of other Previous Vermissers. You've just proving a very different medium coming out of the Permisser elementary and middle behinds, a much brighter medium.

As proof, the prime to her fee

-

vio fire and very on so continue receive at MLT, Georgeover and Very Price. The educations that there are marken at both Provon fire and Province Very who are reflacing conductably has marken that their problems are often the first of the schools review them.

Teching men been, more entraine, the releast day must be largelmed, the community ment provides
may use references in movemen.
Toking of his on how compains
that we have us track the may. "The
more things that the house hists are
most on he housed his are
most on he housed his are

"It's not that your didd is bester than my child," says Jusies Bessers, where draphers, Christ, seeks in the up 10 parases of the justice often as Province West, "It's only order than

But the approxime is becoming wider of the time. Provide West is the cample of the priofel dearns of non-time is being played out in many communities. A hird of segmine system, which is supposed to integration by describing their communities the measure blocks make in. Blocks, suching subsubstantially just as the crime and dearn of the city just as which of 30 years ordine, more, substantial of the city just as which of the day just as and or find the contract ordine.

To PREAT, JAN TO MAI. NATcrees, a Provine West senial
sender teacher, it's heart a long
sender teacher, it's heart a long
weak. The impienters detent's
come to coully separate, and it is on
days file this, when he feels treed, just
weating to go heart, that the VCR
come is to hearly. It's the sentent
wides of the weak for his sentently;
cheese. The files was a "MAII" sepcome on an occasion enclosed by a
lamme Cachelic priors, and Presents
the chigal to weat the sentently
in which the subject of the convenien
sentent and speaks in tempera and
has to be deal to a chair, might be une
very to weath. Receiptly 10 common
and the files gave being weathed, the
present of human superfection is
shown. Presents hand and ascele to
her, Presents hand a speak to.

"Here many people bears show the from pursual experiences" he also the date, which is made up of points and content and is mostly points. The content follows in

THE NEW YORK THUS MARAGINE / Har 20, 1994 29

hard Any mashes?" Some of the mashese heigh emberrancelly, but then, the is Mr. Process, or Mr. P. is he renders call him, and Mr. P.

With the film playing Pooran common his firm of successing, "Ve per seporan prognets in here" She never her hard and looks at his above quintiently, as if trying to figure out what on earth reach a quantum his to do with assistings about the says arthing, hoping to disness the whole thing with a look hopply, but he doorn't he hash hopply, but he doorn't he

There are dops, percentarly with his justice because char, when Pressuen is an electric mether. In trying to explain the thereates behind the French Revolution, for example, he used to set revoke to execute what it would be the if there were a revolution at Province Wee.

statum rain, preducing a first, quirted data. He down't do much of that superiors, and he captains why. I have run into problems now dry. I have a her of black high. They

Over the part 20 years, the school's readonic performance he februal. Some tenthers say that are done in general an integer laws there for humany of the orbit age and the immediacy of the orbit age and the video age and are to beyond the sideo age and are the video age and are to beyond the sideo age and the same 24 hours. These tenthers do say that, given the changing consense is landscape, it's difficult to blant mediane. It must be hard, they say for a Provine West readone to the sad and see compay second to the sad and see compay second to the following or to have reasons from older friends who have centre best home with their backward only pay the room, much have expect and pay the room, much have expect and pay the room. The breaks my hour when I think about when they're going to be it 10 years."

Reballo is also assessly source of the

Provine West stackers purely on the basis of chair stake. Upon bearing that the bad subm a jub at a subset that was more than 30 purers; mining, sequences install purers; mining, sequences in a lad subset with had high." They hape suffing her than the only way so survive would be with sunglesses and signifier. Due when the gast to Provine West has fell at a new subset, Robello found sendence who was open, pusating for standard was well hare," of the same.

Pearson, who is 50 and served at the school in 1971 when it was virtually all whim, one the place very differently. In relating what he believe is the increasing what he believe is the increase the services—assist changes that have transferently the sevently over the part 20 years have also ferouse on one said group — the blacks. His connection of a person said among some whim madeus. There are stone that madeus, if don't want to see asyone black," he says. The just hel it."

Parrier des behavier de

3

blacks as the halfways horsean almost
(The constant of some of shoot
black scales a convolidly assumes—
the yalfan, the servenency, the very
they his girls.) He with about the
time show blacks came use his date
and search become up at the (As is
black scales with a bin. (As is
begannel, now of the three seasonal
the school.) He with season of white
the school.) He with season of white
the school.) He with states of white
these day couldn't season being with
blacks. (You have, life, P., Pin jinling us as English IV Academic chas,
and I'm saving have with these black
tick who out i oven were a complete
tenses."). He also with the sway of
fections tracker, now servered, who
tracked so a small amount ideal on be
tracked so a small amount ideal on be

"I wish I send have the year," up: Persons. "There are days I make how when I are hardly four is. I have no foure agreed."

Parviso war was comparviso war an of comand read sphered in Americ
at one. Bude way manag or Change from the Small by the
demands, and when respected by
pring one or feet on they would. Many
of these where not dury would no
from Toronthy because of impresently to the man, as strong one
states area, as relatedly shop homing and in absence of blacks.

Danag men of the 1969's, Province Was was authoristly when and Prosign Ben productionally whice, Per ages of the details the schools was mith. The security we mith. It on Day 4, 1988, Faul Hangara, of history disc Chingo Blath Fundament that the Chingo Blath Fundament mail on the cap's We Mayarad, where Provide fact to the cabes with the cape of Day 18, other white makes appeared a blath-creater species opposed a blath-creater species of the cape and the cape of the cape of

The size shock the country to in our. White Eight began opin, are in the conditional parties have the size on the substant parties have the size of an about, but from Process in Mayoraci, these tree \$2.20 white and \$2,00 kinds, by \$100, the white count had dragand evolution and \$2,00 kinds, by \$100, the white count had dragand evolution are discussed but had been block parties out of \$2,70 in \$100, Techy, Milwood had one block parties out of \$2,70 in \$100, Techy, Milwood are discussed to draganeous in Process Termship my it happened the way it \$61 on the Parties of Chinas Sales of the way it \$64 on the Parties of Chinas Sales of the way it \$64 on the Parties of the minute in labels in the sales of the minute in the sales of the minute in the sales of the way it \$64 on the parties of the minute in the sales of the minute in the sales of th

For a sine, Novice Was High School consisted involved from the constituted from the constituted from the constituted from the constituted from the first of the Property of the constituted from the constituted from the first from the first from the Property of the constituted from the first from the fi

"The implication was it was 't point so be pressy," upo Busses. There was the feeling as the conmentry that Provide Vere "verse" built for black bulk," she cope "Black bids with white lasts was going to man the observes system over there." So the white the device white colors to when the bessegment of their convenients is threement. The many ways, they show decard the obset."

Des compand extends an bissens in the subsets, and so an exceptface's Out Park, part a count's down term Research

In the law 1900, a second submentages predicted that passe of Oak Path, which between visually of black, would become visually of black by 1908. He count of decrease up their bands and floring for the subside, resident phased for composite and volument is. "It was a commany-wish offers where the second country was a deciman to the second country was a second of subsection of a second of country as a second of subsection of a second of second of country of a second of subsection of a second of second of country as a second of country of second labels of Oak Path and Birst Passes High School. To provide of which he had been so employ to the last of the second second second labels of the last beautiful or subsection of the last beautiful or subsection of the second second second labels and their beautiful of Chings which he had been so employed the beautiful or subsection of the second se

To entrange a state of stability, the village provisioner proved an ordered provision. See of provisions provided approximate the provisions of the provisio

Ber Od Polt is an arrapson. These some or deing soding to insure from lappening the deserve-

THE POW TODE THESE MASSAGE! MAY 30, 1904 30

ove presses of erral dange det he subset place in the county count," my Gary Orficht, a professor of others are and county pating as Harmord and the author of a count seasy showing as arminesed word toward agraph com at the general subsets are the last of famines of agraes of the pating of agraes of the county of the

His rendy, conducted for the November School Branch Assessment showed that as the subjects around the largest cines, 35 persons of block and 64 persons of Lancent around schools that are at least half canoning. Given that existency conditions of rapidly statutoring in these subjects arranged varieties of programment of the state of the state

"The phone story are marriag or well go diversify the same present of 10 to 15 years."

ONTILAT TO PROVIDE Wat's represent for values and revent, often forward by people who have see from sands during school tears, the pitch is excludy and quantities and patients, offering security and quantities of public advances, is expenses and gamera, offering security seed specifically as the security of public advances, is expenses and gamera, offering security security of the values of the security of the

It is true that A.C.T. seems at brown. West are below those of severating reburban discusse. It is not that the number of students that are high as in those discuss. But it is high as in those discuss. But it is the over that many Province sertants have come from the difficultions have come from the difficultions have come from the difficultions are of Channe, and that own witing their new continues of the companions of the companions of their safe companions. Many also continues for facilities are produced to their particular and p

We are stored the mean complex store in the heaver of the desire," says There is junden. Provine Week present. When he came to the school in 1992, he says a place that confirmed that fore-place impresses. If the forest to the

as a fire assempt the development option of the school. The assemble reveal assemble, he felt, was specifical by the lack of doors as the hadrons assemble, over the sufficient.

"It's our very departy stand," says Jacks. "It remain very simple but observed and the best standard of stades here govern for spect," And for the part two years he he worked to bridge this gap. He per doors back on the hedronaux. He stadished the requirement for stades to very ID's averand that such the large way. As the state time, he has been trough, attentions a street way, attentions a street way, attentions a street way.

But he she know that the read to reservey for Provise. West at long and he speaks blandy cheer on terms of the matter of a school base of the caseline of a school base for 20 years, during the most provide size in the horsey of the Provise disease, femal security over disease, femal security over diseases; and not cannot provide cheeren; and not cannot provide otherwise; and the cannot be improved to the special provide of the special provide of the form in the disease of the special provide of the special provides of

White may not be randing that children to Fronce West in significant members represent, but they all cost exercises and Fronce Ers. This behind discriming the a very high reduced discriming and produced and the a very high proportion of discript resistant. If descript they retain to supplies the stands of the proportion of discript resistant. If descript they retain the stands in appears proporty-month in the stands of appears proporty-month them to appear proporty-month in the stands of the produced and the supplies are being an according to the produced and the stands.

Saint 1981, there have been devereferentions enting versus to approve increases in the property tot ran for Provine Township High Subsed Dissert 2007; otherwise fanel, and each his been defenred. Schools at compayer for come enterty of the size, bre in the Provine destire the cost is death-orbital to minimal the cost of the Provine destire the cost of the Provine for the cost of the Provine West with sevone bearing problems continues to grow, and other increases on the the cost of 20 persons of all students at the school on "it sink" analysische. "You wast to strench and propert and help get ready these resolution who are Conford, the distract's assessment of presentation for convenient and of convenient. They need more assessing. The resources they need of openitionally more than for the involve is O.E."

Financial relief scenes unificate harvers, eight at the head or stand harver, eight at the head or stand harver, eight at the head or stand about in the stand on often utilised, but the next result of the impartituded features. The standard becomes manager, has ease last. Standard becomes another the familiar budget in 1994 to though 7 persons for the family year that each in Jane. There always have appelled at the way in which the standard feature in family at the last affects of Court, here presented observed in this age," any Parella, who is resting this year and feats for a most his saint.

If the State Laguebrane does extend and load employee do earling the early other chains is obvious. Continuous gas are no the bone, assumed a substitute gas set to the bone, resulted environmental acceptance for female West, to provide a comprehensive administration for everyone, because absences for everyone, because absences for everyone, because

ion to obsained spirales of a public bracing project. These web the process could stay become they have so other phase to grant annual fire further were to other subsaints, or to the provided and private schools that sing Provided The Little because.

Too do a debendhen on." spe Janus Royd, a school band make. Thes do remains, can also do con on onio a loss.

In the monetion, some of these who work in the distain continue to light the good light. A new representation was knot for the distaint in 1992 after a mineral result. New principals were put in place as Province West and Province Lim. The released bound has been recommend. A constally wiston resonance the large distaint and a temporal content of moneting unity and a manner access of moneting.

Process Street desert here a problem with observed order. In the oppositence of a marker of Persian Ware since 1900, the her seem down own old go. On paper they remain weatheful, remeries improved deserving feet period, they seem as ward on her period, they seem as ward on the period of the Jackson with a book in his hands. Man of the suchase in her remedial Raylish class or bushesse who, though our typical of the surjective of Province West suchases, and as the third-peak to the fifth-grade level, assering Rivers to reache when, if ampling, they were emple before they get here. "It was been under the feet them, and as the first them." It was been as the feet them to be at a marriary level to get a good job." A pid made on a tellipsy. A bry suchase his finger.

Today, Navar's day and acreba remedial English days are weeking register. The assignment is to take a summent by a second-black figure and caption the constant of its memory. Two planning banks of Manimodcomponent fleak the with. The secdense in draws before them. There is the familiar white and didn't of flappy dish into dish drive. Some just more at the servers. Some play, pling up days window after window on the planning stream files think layous of pain, so the point when they can not large find the fits they wan weakters in the Comment on some

THE NEW YORK TIMES MAKASING / May 30, 1904 25

High School

feet plans. Seeme verter a neumans, maybe vers, before becoming developing bedoed up, become, defining from eich ver eich, milling for der werden become darp mend belgs, derpossely pand belgs.

"Miss Bares ... Hop, Miss Bares ..."

She works the steem dispurely, pairing and providing and marrowing and experiening, assering leads and furelfrom Marrowsh to Marioman She knows the differenon eryong, and the in eryong. Shortly, his by his, words uppure on the stronger

The papel from our Morin Later Ring Jr. Loud did not only get along Moren Later Ring warned wagb and blace pusple.

Dr. Moren Later Ring fr. Broph the word in size and some of my franch in. The far present to a late, to do but day on his so late. By cutting her and present 5 care as popular hair. Life the other day, when I was backing at TV the deer half rang, and my little involvement in. It was done large with some rang and my later to large range my later. So I am there have, because they need it. And they wave large.

Wedness on America is very bad. I now a lot of bad denig on my life, was I now a long get dire as a few when I load as a was about three people get deat deat night. Our time I now dea was get heat no load deat for was orying bland.

in the last of the

I her eight and who did not the in for and and who death they are falle the corp through a question to planting the is way to be a second and Conserved the William have been any language. They do th

Till on ay your soon to enhance you I said your

And now the is of threigh the classroom, and so one is rafe, not own the girl at the beat who he calm her pusifier out of it. Such a phone belief out of it. Such a phone belief out it. Such a proving it," she say the such as the such the state of the state of the state of the state. Write all learning." Her statemen in the class, repleases in a callege proportion in the class, represently consistent, and can thing it for reast name of the statement have their heads reast in the statement of the

So find her primary vision. It is Ecith, She reads a pumpe from "Julius Conces," Act I, Sense S: "Why, there was a course effected him and hing offered him, he put is by with the hand of his hand, then, and then the pumple full date, and then the pumple full contents."

See sales faith as explain. He says arching See repeate the present of these from leader, so that present each receive on all three flowers of the school can have it. He miss so say he drought andonesed is, but the case him off side contents of the content of t

Yearn as a salesia Yearn appeard to lost as I.O. arm MT

The read contact printings, and come upon these or the come officer.

"Are you enjoy! Are you

He sell plack ignorma, like is described to process. After the part of the process and the process are process and the process and the process are process are process and the process are process are process and the process are process are process are process are process and the process are pro

"I can oping to open up the parage increase the case of other case of the case of the case opined with a familial transport, and the case opined with a familial transport, and the case opined to the case of the

"We have some unchars
who my there high ear't leave
and deart leave this and
deart leave then," mys Welhan. "I think some machine
are so as ensure, This as
Province," and they deart leave
to ensure."

HICH SCHOOL Comment from page 45

Presently by saying their. When the year the shift Brenty distance may a restaur who is finding a and the heave," my Suphia the read, people beyong to search, barrer that he can Departs, a Province East Eng-hear the appearant. To has do bester, presents to do life tender who adventue a

fail, a's use had. Failure a she

TALLACE MAS

"Gro presentine a band"

in the selection of the control of t

High School.

Hamiltonian mines

general medical selection

for the selection of the sel

American Indian company of the compa

As the do entire bath

I denote to saled day

The little of the or

The probability of the or

The little of t

Thy did by an

Town and to an Reservice area of Strategy

Top one of

ause count a don for side of these descring it is our form?

When what is death of spiral black, one of the coman spiral black one Toman and hard

One makes lating up a man in 2000 A Spee On years' what a group of a per grow out of manch "20 a gried day in the latings, days when it hash the lates" Others lands and and data latest approximately. How data lates approximately. How data lates approximately. How data lates approximately. How data lates approximately.

Baylo 66 crospless
day 6 ages des day les
princes a comp charles a
Jordo Wen, Juy de les
a principal calculus for filia
des calcults publices as
des calcults publices as
des calcults publices as
des calcults constant
des calcults as calculus
des calcults in calculus
des calculus
des calcults in calculus
des

Transcellent of the country of the c

into was plad to leas integrito orbed yes, di d into wide. And wise t integrandi indo es, i antegrandi indo es, i

When they call themselves, is instructed blammer, down and the second and the sec

Then their content de off deep de veg color their colors des con des, dess tors deg co constitution to the color constitution to the colors on the beam made. "My or you in a day with all the state of the state

Immediately make the surpress of the final field of the first state of

the day makes to be for it do about and day as an analy of dady posses. Just a standard to be about and day of day posses. Just a standard day of day between the day and a standard day on his day on his day of day and day on his day of Just a standard day of the da

Amount of the same of children and the same in the sam

despites a parameter bring remeding of an executy at the related, has a desay's bother has a whole lot a

The beart said our so Whitey," says James Bearers, "The's cald our so har come, who will like he hear." Through her despites, she of the every of her come white cardens are dispressed and every to have. But the bearest do bearest deliberated and every the heart of the publish and common despites, has become of the publish and common despites, has become of the publish and common despites, has become of the publish and common despites, and, "and it's just Hispanian and blacks, do quality of the school off go down."

others developer in the second school developer. The median issue print, SUSTAN, is care than SUARE higher than that of Belleval, Only down 5 persons of Province West's resolutes come from Westhernes, arrows for despectly when, Westhernes arrows for despectly when, Westhernes arrows for despectly down on the well with energy who live there, and despectly down on the well with energy who live there, and despectly down on the well with energy who live there, and despectly down on the well with energy who live there, and despectly down on the well with the part of reddinging Westhernes with a high school is outpillered to Levene Township.

Lyons Township High School did parless been audomically thus Provine Was in coveral bay area, he do contain that executed the case attention had so do with raind breakdown. As the time documentains we brong document, Provine Was was 3° parameterism. They would all one and say, I don't wast any white daughor so no so cheed with these blacks over them, " rough Robert Miless, the former principal of Provine Was, " But the statement of plant and their according to years, " I don't statement of the former principal of Provine Was, " But the statement of the former principal of Provine Was, " But the statement of the former principal of Provine Was, " But the statement of the former parameters, they said, "Oh, on, of

Her David Ricarderi, a Warehaster village strates who spearlanded the convenent, calls the charge of ration a convenient reache strong for the relative season and reaches strate of the related. The chart is to long their has an son, there a very convenient for the strate of do-convenient is frequently logical and procedural compliantiess, is ded without cough program. For Ricarderi caps he thinks angustee feelings shout Provide Was have only strongled enough these who live in Warehaster becomes "they think it's a last course, that is angust to correct revenue."

BLUCTANTLY, ALMOST PARSULLY, As and Study Lady would probably upon. Use the rates Western attents who say increases the wholes of delicing in the Lady would prove the lady would prove the lady would be to delice to make a securior to Western They would to make a securior to Western residence, so say that there is a desire in Provint West, a good above. The lad provided school derve my threat of my En," say, Sody Lady. "We would see him to be in a bigger any commonst. We work so maked above than being a continuous. We were so maked above than being a continuous.

They private the parent-reader experience of the school, but the comber of parents yet could count of ... 35 to 45 cas of a content paperhalos of course data. 2.05 ... serve content. And a street data, of 5 cases, that denies are content orders, and offer the counter content and offer the counter content and offer the counter counter counter and offer the counter counter the counter counter counter the counter counter counter the counter counter counter the counter co

or at Provide Week, and they are vising for her so graduar so as Al Lasty pass at "We're

To fought the our and our the ups. "That's how I feel. If

"You win't mer out as most," mer fine. "Ve'n all most neve."

THE END OF ANOTHER He

dry is Provinc West, and for ing the houndary changes in Ba-Dinnes Bests, a sensal reader rope during World Wes L. Over smother, a's not a measure two smother blackboard burgs a picky sense. He came to the school in sussement that reads, "Not Pro-

Toront der von H milien bel is de milesy a Verli V.

"Buy George" ups a mades.
"All sight, we'll may have Mon-day, I wan't be hare. We'll have a

The bid sings and the sends walks are of chan.

Appendix II Letters from the Commission to the Attorney General and Secretary of Education

June 6, 1994

The Honorable Janer Reno Attorney General U.S. Department of Justice 10th Street and Constitution Avenue, NW Washington, D.C. 20530

Dear Ms. Reno:

The U.S. Commission on Civil Rights views with great concern the situation prevailing in the suburban Chicago Proviso Township High Schools District 209, as described in recent media reports (see the enclosed article from *The New York Times*, May 29, 1994).

The racial attitudes and inappropriate classroom behavior ascribed to certain staff, if true, may combine to deprive minority students of their constitutional right to equal educational opportunity. The circumstances reported in the media seem to warrant in-depth review and investigation by your office to determine whether the district has violated the Civil Rights Act of 1964 and other laws barriag discrimination in public schools and in fluorally assisted programs. We urge you to undertake an investigation and inform us as to the results of your inquiry, including any corrective measures you may propose.

I look forward to hearing from you.

· Michay donny

Sincerely.

FOR THE COMMISSION,

MARY FRANCES BERRY

Chairperson

Enclosure

June 6, 1994

The Honorable Richard Riley Secretary of Education U.S. Department of Education 400 Maryland Avenue, SW Washington, D.C. 20202

Dear Mr. Secretary:

The U.S. Commission on Civil Rights views with great concern the situation prevailing in the suburban Chicago Proviso Township High Schools District 209, as described in recent media reports (see the enclosed article from The New York Times, May 29, 1994).

The racial attitudes and inappropriate classroom behavior ancilled to carmin stuff, if true, may combine to deprive minority students of their constitutional right to equal educational opportunity. The circumstances reported in the media seem to wazzant in-depth seview and investigation by your office to determine whether the district has violated the Civil Rights Act of 1964 and other laws barring discrimination in public schools and in federally assisted programs. We urge you to undertake an investigation and inform us as to the results of your inquiry, including any corrective measures you may propose.

I look forward to hearing from you.

e Marie, alere aglicione

Sincerely.

FOR THE COMMISSION.

MARY FRANCES BERRY

Chairperson

Enclosure

Appendix III Letter from Jeff VanderMolen to the New York Times Magazine, published June 26, 1994

As a student at Proviso West and co-editor-in-chief of our school paper, I found your article one-sided. With such a racially and ethnically diverse student body and faculty. I have gotten to know different cultures firsthand, something you cannot get from a classroom lecture.

In your article, Bill Paterson was depicted as a racist. I had a clear with Mr. P., as we call him, for the past year, and not once have I seen that side of him. The man I've come to admire has participated in marches for equal rights, helped fight for voting rights and stood up for blacks in a recent class debate on racism in America.

Jeff VanderMolen Westchester, Ill.

Appendix IV Faculty Institute Agenda, Proviso Township High Schools, August 25, 1994

General Session: All Certified and Noncertificated Staff
Welcome and Introductions8:00 a.m. Dale R. Crawford, Assistant Superintendent
Opening Address8:15 a.m. "Serving All Students Well" Eric L. Eversley, Ed.D., Superintendent
Synergy of Others9:00 a.m. Cultural Diversity in the Workplace, SST Communications
Breakout Sessions: Certified Staff
Breakout Sessions: Round #110:10 a.m.
The School Improvement Plan
Graduation Requirements and Outcomes
A World of Difference
Technology: Where We Are and Where We Want To Be
Quality Relationships: Students and Staff
Project Prometheus
Conversation with the Superintendent
After the New York Times Magazine: "Sorting Out the Issues"
Collaborative Teaching
Building Cross-Cultural Competence
Effective Time Management for Teachers
Lunch
Breakout Sessions: Round #2
Breakout Sessions: Round #3 2:00 p.m. Morning Breakouts Repeated
Dismissal at 3:20 p.m.

A WORLD OF DIFFERENCE

A World of Difference is a training program designed to help teachers combat bigotry and racism while encouraging understanding and respect among racial, religious, and ethnic groups in the classroom. through an interactive approach, this session will help teachers address diversity in the classroom, examine their own and others' biases, and expand their own and their students' cultural awareness.

AFTER THE NEW YORK TIMES MAGAZINE: "SORTING OUT THE ISSUES"

A facilitator will work with participants to identify and strategize around the issues raised by the May 29, 1994, New York Times Magazine. Participants will discuss ways to confront the Times article in a variety of contexts -- teacher to teacher, teacher to student, student to student, teacher to parent -- in an effort to strengthen our abilities as individuals and as members of an organization to work effectively with cultural diversity.

(NOTE: Participants may remain in this session as long as they wish or join the session during any breakout; this session will be the basis for an on-going program during 1994-95.)

BUILDING CROSS CULTURAL COMPETENCE

What are the knowledge, skills, and attitudes teachers need to promote the highest levels of academic success among students of all cultures? This session answers that question by examining ten themes emerging from educational research in learning in multicultural settings. Among the issues highlighted are communication techniques, instructional strategies, and varied learning styles.

(NOTE: This session will be the basis for an on-going staff development program during the 1994-95 school year.)

Appendix V Response of H.G. Bissinger

H.G. MESMGER II

7914 Ardieigh & Phile., PA 19118

July 26, 1986

Constance Davis, Regional Director United States Commission on Civil Rights Michrestorn Regional Office Xerox Centre, Ste. 410 55 West Monroe St. Chicago, IL 60803

Door Ms. Devic:

Enclosed is my written response to the draft report. Given the tener of the testimony of the witnesses, I hope that my response be printed in full. It seems only fair given the standarous attack that was heaped upon me for telling the truth. If there are any questions I can be reached at 215-247-4721.

Sincerely,

H.G. Bissinger M

[&]quot;Personnet to Commission administrative pressdure, a draft of the report, "Rase Relations and Equal Education Opportunity at Previous West High School," was previded to H.G. Biomisgor, author of the article, "We're All Regist Hom." His response was received by the Midwestern Regional Office on July 28, 2006, and is presented in its cationty.

RESPONSE TO THE UNITED STATES COMMISSION ON CIVIL RIGHTS: RACE RELATIONS AND EQUAL EDUCATION OPPORTUNITY AT PROVISO WEST HIGH SCHOOL:

By H.G. Bissinger

When I learned that the United States Commission on Civil rights would be conducting hearings at Proviso West High School based upon the article I wrote for the New York Times, my first reaction was one of gratification. I had spent nearly two months at the school in 1994 in preparation for the article, and I came away convinced that the racial atmosphere existing there was not isolated but representative of the country as a whole. But the more I thought about the hearings, the more trepidation I felt. I have been a reporter for 20 years, and I know by now the inevitable community reaction to reportage that is incisive and honest to the bone. Instead of any real contemplation or serious wish to address the issues, what invariably follows is a kind of mob action to kill the messenger.

After reading your report, Race Relations and Equal Education Opportunity at Proviso West High School, my worst fears have been confirmed. What could have been a serious and urgently needed discussion largely digressed into a slanderous witch hunt against myself and the New York Times Magazine. As I say, I am not surprised by that reaction, only saddened by it. I also want to make something emphatically clear:

I stand by every single word in this stary, not 95 percent, or 98 percent, but 100 percent. Every single word of what was written was honestly reported. No one was misquoted. Nothing, absolutely nothing, was taken out of context. Nothing was embellished or heightened.

This story was subjected to my own rigorous standards as a reporter, which includes a Pulitzer Prize for investigative reporting, a Nieman fellowship at Harvard, and a best-selling non-fiction book. It was also subjected to the thorough and rigorous editing and factchecking process of the New York Times Magazine.

While I applaud the efforts of the Illinois Advisory Committee in addressing the crucial issue of racial attitudes at an American high school, I cannot help but question the wisdom of thinking that the very private matter of race could ever be genuinely discussed in the forum of a public hearing. In my experience not simply as a reporter but as a member of the human race, it is very difficult to get anyone to talk honestly about race in private, much less in public with peers, colleagues and the media zeroing in on every word.

So why was the reaction to the story so extreme? Because certain people within the story got caught. Caught at what? Caught at speaking honestly about the subject of race without any awareness of the hideousness and repugnance of their comments. What happened after they were exposed? They reacted the way most of us would, by trying to deflect their shame onto someone else. In my career as a reporter, I have never been through a more disturbing and shocking experience than that Winter Afternoon in 1994 when I interviewed six white honor students at Proviso West High School. I purposely picked these students because they represented the academic best and brightest of Proviso.

By virtue of their intellectual enlightenment, I also thought they might be less prone to racial stereotypes. Here is a taste of what they had to say about blacks as quoted in the article:

I cannot stand the race. I'll never date anyone who isn't white.

They're always saying 'slave' this and 'slave' that. Sorry, I don't know a slave. I never owned a slave. I'm sick of them throwing it in our face.

Ignorant and scum, a lot of poverty, self-righteous, you owe me that, you owe me this, gimme, gimme, gimme.

In reading the testimony of those who appeared before the committee, comments such as these were dismissed as being incidental and unimportant. If those dismissals didn't work, then Plan B went in effect and I as the reporter was accused of putting words into people's mouths or making up quotes altogether. With the laudable exception of superintendent Eric Eversley, person after person who came before the committee bent over to an almost absurd degree to describe Proviso West as a perfect and harmonious place. That simply was not the case during the period in 1994 in which I was there. The attitudes of various students and teachers that I accurately quoted did not reflect that. Nor did the demographic data: In 1973-74, Proviso West High School was 98 percent white. Twenty years later the school was 18 percent white, despite the fact that whites were still the predominant racial group in Proviso Township. What does that mean? We can paint pretty pictures and be politically correct, but the answer is obvious: As blacks and Latinos started coming into the high school, white parents started pulling their children out.

Does that make Proviso West High School somehow unique? Of course it doesn't. The racial transformation that has taken place there has taken place at thousands of high school across this country. Just look at our urban schools. And as the article shows, it is now time we start looking at our suburban ones.

What is that racial transformation caused by? The experience of Proviso West, if it tells us anything, tells us that it caused by fear, misunuerstanding, resentment, and in some cases outright contempt. I did not write this article with the intention of hurting or embarrassing anyone. But I did write it with the intention of telling the truth, however painful, because the parable of Proviso West is the parable of our modern America. If people in the community wish to respond with outrage and denial, that is their right. But I will not take back a single word, because I am proud of what I wrote.

